UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, DC 20549

FORM 10-Q

	RLY REPORT PURSUAN GE ACT OF 1934	T TO SECTION 13 OR	15 (d) OF THE SECURITIE	ES
For the qua	rterly period ende	d July 31, 2006		
		OR		
_	TION REPORT PURSUA GE ACT OF 1934	NT TO SECTION 13 O	R 15 (d) OF THE SECURIT	IES
For the tra	nsition period fro	m	to	
Commission	File Number 0-1818	3		
		-III APPAREL GROUP egistrant as speci	, LTD. fied in its charter)	
	Delaware		41-1!	590959
	or other jurisdic oration or organiz		(I.R.S. Identifica	Employer ation No.)
	h Avenue, New York Principal Executi	-)18 Code)
		(212) 403-0500		
	(Registrant's t		ncluding area code)	
	· · · · · · · · · · · · · · · · · · ·	ormer address and changed since last	former fiscal year, report)	
to be filed the precedi required to	by Section 13 or ng 12 months (or f	15(d) of the Securior such shorter per), and (2) has been	1) has filed all reports ities Exchange Act of 19 riod that the registrant n subject to such filing	934 during was
		Yes [X] No [_]		
accelerated	filer or a non-ac	celerated filer. Se	s a large accelerated fi ee definition of "accele 2 of the Exchange Act.	
Large Accel	erated Filer [_]	Accelerated File:	r [_] Non-Accelerated	Filer [X]
	check mark whethe		s a shell company (as de	efined in
		Yes [] No [X]		
As of Septe	mber, 2006, there	_	mmon shares outstanding.	
110 01 DOP 00	mbel, Love, enere		or onaros oacocanaring.	
				Page No.
Part I	FINANCIAL INFORMA	TION		
Item 1.	Financial Stateme	nts (Unaudited)		
		dated Balance Sheed July 31, 2005 and	ts - January 31, 2006	3
	Condensed Consoli	dated Statements of	f Operations -	

	For the Three Months Ended July 31, 2006 and 2005	4
	Condensed Consolidated Statements of Operations - For the Six Months Ended July 31, 2006 and 2005	5
	Condensed Consolidated Statements of Cash Flows - For the Six Months Ended July 31, 2006 and 2005	6
	Notes to Condensed Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	14
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	19
Item 4.	Controls and Procedures	19
Part II	OTHER INFORMATION	
Item 1A.	Risk Factors	20
Item 4.	Submission of Matters to a Vote of Stockholders	20
Item 6.	Exhibits	21

2

PART I

ITEM 1. FINANCIAL STATEMENTS

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except share and per share amounts)

	JULY 31, 2006	July 31, 2005	January 31, 2006
		(unaudited)	
ASSETS			
CURRENT ASSETS			
Cash and cash equivalents Accounts receivable, net of allowance for doubtful	\$ 728	\$ 1,194	\$ 7,031
accounts and sales discounts of \$8,214, \$4,211 and \$9,443, respectively	58 301	53,078	45 751
Inventories, net	81,163	72,727	,
Prepaid and refundable income taxes	7,455	3,951	
Deferred income taxes	4,101	3,357	4,101
Prepaid expenses and other current assets	13,838		
Total current assets	165,586	145,501	95,122
PROPERTY, PLANT AND EQUIPMENT, NET	4,255	3,581	4,296
DEFERRED INCOME TAXES	2,430	2,050	2,415
GOODWILL	18,787		18,501
OTHER INTANGIBLES, NET	13,628	29,288	15,287
OTHER ASSETS	2,468	4,201	2,090
	\$207,154	5184.621	
	======	\$184,621 ======	=======
LIABILITIES AND STOCKHOLDERS' EQUITY			
CURRENT LIABILITIES			
Notes payable	\$ 48,610	\$ 52,206	\$ 7,370
Current maturities of obligations under capital leases	208	151	208
Income taxes payable			2,269
Accounts payable	44,708	31,336	9,749
Contingent purchase price payable			3,380
Accrued expenses	7,340	7,688	10,949
Total current liabilities	100,866	91,381	
NOTES PAYABLE	18,450	25,050	
OTHER NON-CURRENT LIABILITIES	527	774	631
TOTAL LIABILITIES		117,205	

COMMITMENTS AND CONTINGENCIES			
STOCKHOLDERS' EQUITY			
Preferred stock, 1,000,000 shares authorized;			
No shares issued and outstanding in all periods			
Common stock - \$.01 par value; 40,000,000 shares			
authorized; 14,364,125, 12,581,371 and 12,701,222			
shares issued, respectively	144	126	127
Additional paid-in capital	52,132	33,667	36,262
Accumulated other comprehensive income		62	
Retained earnings	36,005	34,531	46,592
	88,281	68,386	82,981
Common stock held in treasury - 367,225 shares at cost	(970)	(970)	(970)
	87,311	67,416	82,011
	\$207,154	\$184,621	\$ 138,317
	=======	=======	=======

The accompanying notes are an integral part of these statements.

3

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except share and per share amounts)

	THRE	E MONTHS E	ENDED	JULY 31,
		(Unaud 2006		
Net sales Cost of goods sold		69,082 52,249		41,804
Gross profit Selling, general and administrative expenses Depreciation and amortization				12,749
Operating (loss) income Interest and financing charges, net		(1,757) 1,264		149 527
Loss before income taxes Income tax benefit				(378) (77)
Net loss		(1,737)		(301)
LOSS PER COMMON SHARE: Basic and Diluted:				
Net loss per common share		(0.14)		(0.03)
Weighted average number of shares outstanding				

The accompanying notes are an integral part of these statements.

4

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except share and per share amounts)

SIX	MONTHS	ENDED	JULY	31,
	(Una	dited)	
	2006		2005	5
\$	83,471			,320
	65 , 959	9	54,	656

Net sales Cost of goods sold

Gross profit Selling, general and administrative expenses Depreciation and amortization	•	13,664 20,922 781
Operating loss Interest and financing charges, net	(16,502) 1,911	(8,039) 530
Loss before income taxes Income tax benefit	· · ·	(8,569) (3,599)
Net loss	\$ (10,587) =======	\$ (4,970)
LOSS PER COMMON SHARE: Basic and Diluted:		
Net loss per common share	\$ (0.85) ======	\$ (0.45) ======
Weighted average number of shares outstanding	12,410,000	11,007,000

The accompanying notes are an integral part of these statements.

5

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands)

SIX MONTHS ENDED JULY 31,

	SIA MONINS E	ADED CODI SI
		dited) 2005
Carly Clause Course and the carlo in the car		
Cash flows from operating activities Net loss	\$ (10 587)	\$ (4,970)
Adjustments to reconcile net loss to net cash used in	\$ (10,307)	\$ (4,570)
operating activities, net of assets and liabilities		
acquired:		
Depreciation and amortization	2,197	781
Non-cash stock based compensation	175	
Deferred financing charges	484	131
Deferred income taxes	(15)	
Changes in operating assets and liabilities:		
Accounts receivable		(20,160)
Inventories, net		(30,106)
Income taxes, net	(9,724)	
Prepaid expenses and other current assets		(10,799)
Other assets	(256)	
Accounts payable and accrued expenses		16,989
Net cash used in operating activities		(49,146)
Cash flows from investing activities		
Capital expenditures	(499)	(648)
Acquisition of Marvin Richards, net of cash acquired	143	(19,623)
Acquisition of Winlit	(73)	(580)
Contingent purchase price paid	(3,380)	
Net cash used in investing activities	(3,809)	(20,851)
, and the second se		
Cash flows from financing activities		
Increase in notes payable, net	41,240	
Proceeds from term loan		30,000
Repayment of term loan	(3,300)	
Payments for capital lease obligations	(104)	
Proceeds from sale of common stock, net	15,035	
Proceeds from exercise of stock options	323	83
Repayment of terminated credit facility		(12,457)
Proceeds from new credit facility		36,405
Net cash provided by financing activities	53,194	
Effect of exchange rate changes on cash and cash equivalents		12
Net decrease in cash and cash equivalents	(6,303)	(15,380)
Cash and cash equivalents at beginning of period	7,031	16,574
Cash and assh anti-salanta at and affirmation		
Cash and cash equivalents at end of period	Ş 128	\$ 1,194

6

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (CONT'D) (in thousands)

				ED JULY	,	
	 2006		naudited) 2005			
Supplemental disclosures of cash flow information: Cash paid during the period for: Interest		,		545		
Income taxes Supplemental schedule of non-cash investing and financing activities:	1	,879		308	3	
Fair value of shares issued in connection with the Marvin Richards acquisition Debt assumed in connection with the Winlit asset acquisition	\$	356	\$	4,685 6,695		
Detail of the Marvin Richards and Winlit acquisitions: Acquired intangibles Fair value of other assets acquired			·	27,819 29,743	9	
Fair value of total assets acquired Liabilities assumed Common stock issued				57,562 (32,655 (4,685	2 5) 5)	
Net cash paid for acquisitions Cash acquired				20,222	2	
Cash paid for acquisitions			\$	20,203	3	

The accompanying notes are an integral part of these statements.

7

G-III APPAREL GROUP, LTD. AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Note 1 - General Discussion

As used in these financial statements, the term "Company" refers to G-III Apparel Group, Ltd. and its majority-owned subsidiaries. The results for the three and six month periods ended July 31, 2006 are not necessarily indicative of the results expected for the entire fiscal year, given the seasonal nature of the Company's business. The accompanying financial statements included herein are unaudited. In the opinion of management, all adjustments (consisting of only normal recurring adjustments) necessary for a fair presentation of the financial position, results of operations and cash flows for the interim period presented have been reflected.

Certain amounts in the Condensed Consolidated Balance Sheet as of July 31, 2005 and the Condensed Consolidated Statements of Operations for the three and six months ended July 31, 2005 have been reclassified to conform to the current period presentation.

The Company consolidates the accounts of all its wholly-owned subsidiaries. All material intercompany balances and transactions have been eliminated.

All share and per share data have been adjusted to give retroactive effect to a three-for-two split of the Company's Common Stock effected on March 28, 2006.

The accompanying financial statements should be read in conjunction with the financial statements and notes included in the Company's Annual Report on Form

10-K filed with the Securities and Exchange Commission for the year ended January 31, 2006.

Note 2 - Private Placement

On July 13, 2006, the Company completed a private placement of its Common Stock and five-year warrants to purchase its Common Stock. The Company issued 1,500,000 shares of Common Stock at a price of \$10.11 per share, resulting in aggregate proceeds to the Company of \$15,165,000. The Company also issued warrants to purchase an aggregate of up to 375,000 shares of its Common Stock, exercisable beginning six months after the closing date of the private placement, at an exercise price of \$11.00 per share, subject to adjustment upon the occurrence of specified events, including customary weighted average price anti-dilution adjustments. The proceeds were used to repay a portion of the outstanding balance under the Company's revolving credit line.

For two years after the closing date of the private placement, the investors will, subject to exceptions and qualifications specified in the purchase agreement, have a right of first refusal with respect to the proposed sale by the Company of its equity or equity equivalent securities if such sale is at an effective price per share of \$10.00 or less.

The Company also entered into a registration rights agreement with the investors, in which it agreed to file a registration statement with the Securities and Exchange Commission to register under the Securities Act of 1933, as amended, resales from time to time of the shares, any warrant shares issued upon exercise of the warrants and an additional 500,000 shares of Common Stock sold to the investors by Mr. Aron Goldfarb on July 13, 2006. The Company filed the registration statement within the required time period and the registration statement has been declared effective.

Note 3 - Stock Based Compensation

Effective February 1, 2006, the Company adopted Statement of Financial Accounting Standards No. 123R, "Share Based Payment" ("SFAS 123R"). The Company elected to use the modified prospective transition method; therefore, prior period results were not restated. Prior to the adoption of SFAS 123R, stock-based compensation expense related to stock options was not recognized in the results of operations if the exercise price was at least equal to the market value of the common stock on the grant date, in accordance with Accounting Principles Board Opinion No. 25, "Accounting for Stock Issued to Employees." As a result, the

8

recognition of stock-based compensation expense in prior periods was generally limited to the expense attributed to restricted stock awards.

SFAS 123R requires all share-based payments to employees, including grants of employee stock options, to be recognized as compensation expense over the service period (generally the vesting period) in the consolidated financial statements based on their fair values. Under the modified prospective method, awards that were granted, modified, or settled on or after February 1, 2006 are measured and accounted for in accordance with SFAS 123R. Unvested equity-based awards that were granted prior to February 1, 2006 will continue to be accounted for in accordance with SFAS 123, except that all awards are recognized in the results of operations over the remaining vesting periods. The impact of forfeitures that may occur prior to vesting is also estimated and considered in the amount recognized. Also, the realization of tax benefits in excess of amounts recognized for financial reporting purposes will be recognized in the Consolidated Statement of Cash Flows as a financing activity rather than an operating activity as it was classified in the past.

It is the Company's policy to grant stock options at prices not less than the fair market value on the date of the grant. Option terms, vesting and exercise periods vary, except that the term of an option may not exceed ten years.

The following table summarizes the pro forma effect of stock-based compensation as if the fair value method of accounting for stock compensation had been applied for the three and six months ended July 31, 2005.

	Three Months Ended July 31, F	Six Months Ended July 31,
	2005 (in thousands, except p	2005 per share amounts)
Net loss - as reported Deduct: Stock-based employee compensation expense determined under fair value method, net of related tax effects	\$ (301)	\$ (4,970)
	79	154
Pro forma net loss	\$ (380) =====	\$ (5,124) ======
Basic and diluted loss per share - as reported Pro forma basic and diluted loss per share	\$(0.03) \$(0.03)	\$ (0.45) \$ (0.47)

The fair value of stock options was estimated using the Black-Scholes option-pricing model. This model requires the input of subjective assumptions that will usually have a significant impact on the fair value estimate. The assumptions for the current period grants were developed based on SFAS 123R and Securities and Exchange Commission guidance contained in Staff Accounting Bulletin (SAB) No. 107, "Share-Based Payment." The following table summarizes the assumptions used to compute the weighted average fair value of stock option grants.

The following weighted average assumptions were used in the Black-Scholes option pricing model for grants in fiscal 2006 and 2005, respectively:

	2006	2005
Expected stock price volatility	48.7%	67.5%
Expected lives of options		
Directors and officers	7 YEARS	7 years
Employees	6 YEARS	6 years
Risk-free interest rate	3.9%	3.9%
Expected dividend yield	0%	0%

The weighted average volatility for the current period was developed using historical volatility for periods equal to the expected term of the options. An increase in the weighted average volatility assumption will increase stock compensation expense.

9

The risk-free interest rate was developed using the U.S. Treasury yield curve for periods equal to the expected term of the options on the grant date. An increase in the risk-free interest rate will increase stock compensation expense.

The dividend yield is a ratio that estimates the expected dividend payments to shareholders. The Company has not declared a cash dividend and has estimated dividend yield at 0%.

The expected term of stock option grants was developed after considering vesting schedules, life of the option, and historical experience. An increase in the expected holding period will increase stock compensation expense.

SFAS 123R requires the recognition of stock-based compensation for the number of awards that are ultimately expected to vest. As a result, for most awards, recognized stock compensation was reduced for estimated forfeitures prior to vesting primarily based on an historical annual forfeiture rate of 3.6% for stock options. Estimated forfeitures will be reassessed in subsequent periods and may change based on new facts and circumstances.

The following table summarizes stock option activity for the six months ended July 31, 2006:

		Weighted
		average
		exercise
	Shares	price
Options outstanding at beginning of year	1,429,348	\$3.53

Exercised	(162 , 955)	\$2.06
Granted	221,000	\$9.78
Cancelled or forfeited	(122,650)	\$9.18
Options outstanding at end of period	1,364,743	\$4.23
	=======	
Exercisable	942,993	\$3.20
	========	

The weighted average remaining term for stock options outstanding was 4.8 years at July 31, 2006. The aggregate intrinsic value at July 31, 2006 was \$8.6 million for stock options outstanding and \$6.9 million for stock options exercisable. The intrinsic value for stock options is calculated based on the exercise price of the underlying awards and the market price of our common stock as of July 31, 2006, the reporting date.

In connection with an acquisition in July 2005, the Company granted 225,000 shares of common stock subject to vesting based on the future market price of the common stock through January 31, 2009. In August 2005, 37,500 shares vested and in February 2006, an additional 37,500 shares vested as a result of market price conditions being met. The cost for the restricted stock was measured and reflected as additional goodwill based on the quoted market price on the date the shares vested and the restrictions lapsed.

The following table summarizes unvested restricted stock unit activity for the six months ended July 31, 2006:

Unvested	as	of	Febru	ıary	1,	2006	187,5	00
Granted								
Vested							(37,5	00)
Unvested	as	of	July	31,	200	06	150,0	00
							=====	==

10

Proceeds received from the exercise of stock options were approximately \$323,000 and \$83,000 during the six months ended July 31, 2006 and 2005, respectively. The intrinsic value related to the exercise of stock options was \$1.2 million and \$71,000 for the six months ended July 31, 2006 and 2005, respectively, which is currently deductible for tax purposes.

Tax benefits were attributed to the stock-based compensation expense. The Company elected to adopt the alternative method of calculating the historical pool of windfall tax benefits as permitted by FASB Staff Position (FSP) No. SFAS 123R-c, "Transition Election Related to Accounting for the Tax Effects of Share-Based Payment Awards." This is a simplified method to determine the pool of windfall tax benefits that is used in determining the tax effects of stock compensation in the results of operations and cash flow reporting for awards that were outstanding as of the adoption of SFAS 123R.

As of July 31, 2006, approximately \$914,000 of unrecognized stock compensation related to unvested awards (net of estimated forfeitures) is expected to be recognized through July 31, 2010.

Note 4 - New Accounting Pronouncements

In June 2006, the Financial Accounting Standards Board issued FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"). FIN 48 clarifies the accounting for uncertainty in income taxes recognized in the Company's financial statements in accordance with FASB Statement No. 109 "Accounting for Income Taxes." FIN 48 also prescribes a recognition threshold and measurement attribute for the financial statement recognition and measurement of a tax position taken or expected to be taken in a return, as well as guidance on derecognition, classification, interest and penalties and financial statement reporting disclosures. FIN 48 is effective for fiscal years beginning after December 15, 2006. The Company is currently evaluating the requirements and impact of FIN 48 on the Company's consolidated financial statements, and will adopt the provisions on February 1, 2007.

Note 5 - Acquisitions of Marvin Richards and Winlit Group

MARVIN RICHARDS

On July 11, 2005, the Company acquired all of the outstanding capital stock of J. Percy for Marvin Richards, Ltd., all of the membership interests of CK Outerwear, LLC and 50% of the membership interests in Fabio Licensing, LLC, collectively referred to as Marvin Richards. The total consideration paid by the Company in connection with the acquisition of Marvin Richards was \$28.1 million, including associated fees and expenses. The purchase price was allocated to Marvin Richard's assets and liabilities, tangible and intangible (as determined by an independent appraiser), with the excess of the purchase price over the fair value of the net assets acquired of \$15.0 million being recorded as goodwill. The former principals of Marvin Richards are entitled to receive additional purchase price based on the performance of the Company's Marvin Richards business through January 31, 2009. Goodwill will be increased for subsequent earn-out payments based upon performance.

WINLIT

On July 11, 2005, the Company acquired certain operating assets of Winlit Group, Ltd. The total consideration paid by the Company in connection with the acquisition of Winlit was \$8.1 million, including associated fees and expenses. The purchase price was allocated to Winlit's assets and liabilities, tangible and intangible (as determined by an independent appraiser), with the excess of the purchase price over the fair value of the net assets acquired of \$3.8 million being recorded as goodwill. Winlit is entitled to receive additional purchase price based on the performance of the Company's Winlit business through January 31, 2009. Goodwill will be increased for subsequent earn-out payments based upon performance.

The operating results of Marvin Richards and Winlit have been included in the Company's financial statements since July 11, 2005 and are included in the six months ended July 31, 2006. The results of operations from the acquired businesses for the comparable prior year's period do not include the seasonal losses incurred by the acquired companies prior to the date of acquisition.

11

Note 6 - Inventories

Inventories, which are stated at lower of cost (determined by the first-in, first out method) or market, consist of:

	JULY 31, 2006	July 31, 2005	January 31, 2006
		(in thousar	nds)
Finished goods	\$74,331	\$64,624	\$25 , 557
Work-in-process	693	593	80
Raw materials	6,139	7,510	4,758
	\$81,163	\$72 , 727	\$30,395
	======	======	======

Note 7 - Loss per Common Share

Basic loss per share has been computed using the weighted average number of common shares outstanding during each period excluding unvested restricted stock awards that have not met the market condition. Diluted income per share amounts, when applicable, are computed using the weighted average number of common shares and potential dilutive common shares, consisting of stock options and restricted stock, outstanding during the period.

Note 8 - Notes Payable

The Company has a financing agreement with The CIT Group/Commercial Services, Inc., as Agent, for a consortium of banks. The financing agreement, which expires on July 11, 2008, is a senior collateralized credit facility providing for borrowings in the aggregate principal amount of up to \$195 million. The facility consists of a revolving line of credit and a term loan.

The revolving line of credit provides for a maximum line ranging from \$45 million to \$165 million at specific times during the year, provided that there are no borrowings outstanding for at least 45 days during the period from December 1 through April 30 each year. This condition has been met for the current year. Amounts available under the line are subject to borrowing base formulas and over advances as specified in the financing agreement. Borrowings under the line of credit bear interest at the Company's option at the prime rate or LIBOR plus 2.25%.

The term loan in the original principal amount of \$30 million is payable over three years with eleven quarterly installments of principal in the amount of \$1,650,000 and a balloon payment of \$11,850,000 due on July 11, 2008, the maturity date of the loan. Mandatory prepayments are required under the term loan commencing with the fiscal year ending January 31, 2007 to the extent of 50% of excess cash flow, as defined. The term loan bears interest, at the Company's option, at prime plus 1% or LIBOR plus 3.25%.

The financing agreement requires the Company, among other covenants, to maintain certain earnings, tangible net worth and minimum fixed charge coverage ratios as defined. It also limits payments for cash dividends and stock redemption to \$1.5 million plus an additional amount for stock redemptions based on the proceeds of sales of equity securities and limits annual capital expenditures. As of July 31, 2006, we were in compliance with these covenants. The financing agreement is collateralized by all of the assets of the Company.

Notes payable also includes a foreign note payable (\$770,000) by PT Balihides, the Company's inactive Indonesian subsidiary.

Note 9 - Closing of Manufacturing Facility

The unpaid portion of the non-recurring charge associated with the closing of our Indonesian manufacturing facility in December 2002 is included in "Accrued expenses" in the accompanying Consolidated Balance Sheets. The balance in the reserve at July 31, 2006 and January 31, 2006 is \$398,000, and represents accrued expenses and other miscellaneous costs. Based on current estimates, management believes that existing accruals are adequate.

12

Note 10 - Segments

The Company's reportable segments are business units that offer different products and are managed separately. The Company operates in two segments, licensed and non-licensed apparel. The following information is presented for the three and six month periods indicated below:

THREE	MONTHS	ENDED	JULI	31,
2006				2005

	20	006	2005	
	LICENSED	NON- LICENSED	Licensed	Non- Licensed
Net sales	\$42,891	\$26,191	\$25,146	\$29,407
Cost of goods sold	31,897	20,352	19,046	22,758
Gross profit	10,994	5,839	6,100	6,649
Selling, general and administrative	13,493	5,097	8,062	4,538
Operating income (loss)	\$ (2,499)	\$ 742	\$(1,962)	\$ 2,111
	======	=====	======	======

SIX MONTHS ENDED JULY 31,

	2006		2005	
	LICENSED	NON- LICENSED	Licensed	Non- Licensed
Net sales	\$ 56,012	\$27,459	\$36,241	\$32,079
Cost of goods sold	43,134	22,825	29,047	25,609
Gross profit Selling, general and administrative	12,878	4,634	7,194	6,470
	23,510	10,504	14,355	7,348

Operating loss \$ (10,632) \$ (5,870) \$ (7,161) \$ (878)

Included in finished goods inventory at July 31, 2006 are approximately \$51.2 million and \$23.1 million of inventories for licensed and non-licensed apparel, respectively. Included in finished goods at July 31, 2005 are approximately \$37.2 million and \$30.3 million of inventories for licensed and non-licensed apparel, respectively. All other assets are commingled.

Note 11 - Proposed Public Offering

On May 8, 2006, the Company filed a registration statement with the Securities and Exchange Commission for a proposed offering of 4,000,000 shares of its common stock pursuant to which the Company would offer 3,000,000 shares and selling stockholders would offer 1,000,000 shares. The Registration Statement was withdrawn on July 12, 2006.

13

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Unless the context otherwise requires, "G-III", "us", "we" and "our" refer to G-III Apparel Group, Ltd. and its subsidiaries. References to fiscal years refer to the year ended or ending on January 31 of that year.

Statements in this Quarterly Report on Form 10-Q concerning our business outlook or future economic performance; anticipated revenues, expenses or other financial items; product introductions and plans and objectives related thereto; and statements concerning assumptions made or expectations as to any future events, conditions, performance or other matter, are "forward-looking statements" as that term is defined under the Federal securities laws. Forward-looking statements are subject to risks, uncertainties and other factors, which could cause actual results to differ materially from those stated in such statements. Such risks, uncertainties and factors include, but are not limited to, dependence on licensed product, reliance on foreign manufacturers, risks of doing business abroad, the nature of the apparel industry, including changing consumer demand and tastes, seasonality, customer acceptance of new products, the impact of competitive products and pricing, dependence on existing management, possible business disruption from acquisitions, general economic conditions, as well as other risks detailed in the Company's filings with the Securities and Exchange Commission, including this Quarterly Report on Form 10-0.

OVERVIEW

G-III designs, manufactures and markets an extensive range of outerwear and sportswear, including coats, jackets, pants, suits, dresses and other sportswear items under licensed brands, our own proprietary brands and private retail labels. Our products are distributed through a broad mix of retail partners at a variety of price points. The concentration of sales to our largest customers has increased and we expect that our ten largest customers will continue to represent a majority of our sales.

We operate in fashion markets that are highly competitive. Our ability to continuously evaluate and respond to changing consumer demands and tastes, across multiple market segments, distribution channels and geographies, is critical to our success. Although our portfolio of brands is aimed at diversifying our risks in this regard, misjudging shifts in consumer preferences could have a negative effect on our business. Our success in the future will depend on our ability to design products that are accepted in the markets we serve, source the manufacture of our products on a competitive basis, deliver products in a timely manner and continue to diversify our product portfolio and the markets we serve.

We operate our business in two segments, licensed apparel and non-licensed apparel. The licensed apparel segment includes sales of apparel brands licensed by us from third parties. The non-licensed apparel segment includes sales of apparel under private label brands and our own proprietary brands.

The sale of licensed product has been a key element of our business strategy for many years. As part of this strategy, we added new apparel licenses in the past year. We believe that consumers prefer to buy brands they know, and we have continually sought licenses that would increase the portfolio of name brands we can offer through different tiers of retail distribution, for a wider array of products and at a variety of price points.

The operating results of Marvin Richards and Winlit, which we acquired on July 11, 2005, have been included in our financial statements since the date of acquisition. Marvin Richards and Winlit are in the wholesale outerwear business and are subject to the same seasonality that we are. Our results for the first two quarters of fiscal 2006 and for the full 2006 fiscal year exclude the seasonal losses that were incurred by the acquired companies in the first half of fiscal 2006. Results for fiscal 2007 include the full year of operations of the acquired companies, as well as a full year of interest expense and amortization expense relating to the acquisitions. Accordingly, our net losses in the second quarter and first six months of fiscal 2007 were greater than our net losses in the second quarter and first six months of fiscal 2006.

These acquisitions are consistent with our strategy to increase the portfolio of brands that we offer through different tiers of retail distribution. Both transactions are expected to complement our existing group of licensed brands, G-III owned labels and private label programs.

14

We continue to believe that brand owners will look to consolidate the number of licensees they engage to develop product and they will continue to look for licensees with a successful track record of developing brands. We are continually having discussions with licensors regarding new opportunities. It is our objective to continue to expand our product offerings. As a result of our acquisition of Marvin Richards, we have licenses for men's and women's outerwear with Calvin Klein. In September 2005, we entered into a license agreement to manufacture and distribute women's better suits under the Calvin Klein label and in April 2006, we entered into a license agreement to manufacture and distribute women's dresses under the Calvin Klein label. We began shipping the women's suit line in January 2006 and expect to begin shipping women's dresses for holiday 2006. We have had a license agreement with Sean John for men's outerwear for over five years. In March 2006, we added license agreements to manufacture women's sportswear and outerwear under Sean John labels. We expect to launch the Sean John sportswear line in 2007. We also design and produce a line of urban sportswear for Wal-Mart under their Exsto label, which began shipping during the second quarter of fiscal 2007.

Significant trends that are affecting the apparel industry include the continuing consolidation of retail chains, the desire on the part of retailers to consolidate vendors supplying them, the increased focus by department stores on their own private label brands and a shift in consumer shopping preferences away from traditional department stores to other mid-tier and specialty store venues. There has also been continued downward pressure on average retail prices for many categories of apparel. We have responded to these trends by continuing to focus on selling products with recognized brand equity, by attention to design, quality and value and by improving our sourcing capabilities. We believe that our broad distribution capabilities help us to respond to the various shifts by consumers between distribution channels. We also believe that our operational capabilities will enable us to continue to be a vendor of choice for our retail partners.

In June 2006, we entered into a seven-year lease for a distribution center in South Brunswick, New Jersey. This facility contains approximately 305,000 square feet of space which will be used by us for product distribution. Annual rent for this facility is approximately \$1.2 million. As a result of adding this new facility, we will not renew our lease for our distribution center in Edison, NJ which expires in January 2007. The Edison facility contains approximately 89,000 square feet. The additional space is expected to allow us to meet some of our anticipated increased shipping volume. We estimate that the renovation of this new facility will cost us between \$1 million and \$1.5 million and that the facility will be fully operational by May 2007.

In July 2006, we completed the private placement of 1,500,000 shares of our common stock at a price of \$10.11 per share, resulting in net proceeds to us, after expenses related to the placement, of \$15,035,000. We also issued warrants

to the investors to purchase up to 375,000 shares of our common stock at an exercise price of \$11.00 per share, subject to adjustment. We used the net proceeds of this placement to repay a portion of our outstanding credit facility.

RESULTS OF OPERATIONS

THREE MONTHS ENDED JULY 31, 2006 COMPARED TO THREE MONTHS ENDED JULY 31, 2005

Net sales for the three months ended July 31, 2006 increased to \$69.1 million from \$54.6 million in the same period last year. Net sales of licensed apparel increased to \$42.9 million from \$25.1 million, primarily as a result of sales of Calvin Klein suits and outerwear (\$9.0 million). Last year, only a small amount of sales of Calvin Klein outerwear were included in our results for the quarter. Net sales of non-licensed apparel in the three months decreased to \$26.2 million from \$29.4 million, primarily due to a decrease in private label men's outerwear sales (\$6.1 million) offset, in part, by the initial shipments of Exsto private label product (\$1.8 million) which were primarily shipped in July 2006.

Gross profit increased to \$16.8 million, or 24.4% of net sales, for the three month period ended July 31, 2006, from \$12.7 million, or 23.4% of net sales, in the same period last year. The gross profit percentage in our licensed apparel segment was 25.6% in the three month period ended July 31, 2006 compared to 24.3% in the same period last year. Sales of Calvin Klein suits and outerwear resulted in an increase in both margin dollars and our gross profit as a percentage of sales. The gross profit percentage in our non-licensed segment was 22.3% in the three month period ended July 31, 2006 compared to 22.6% in the same period last year.

Selling, general and administrative expenses increased \$6.0 million to \$18.6 million in the three month period ended July 31, 2006 from \$12.6 million in the same period last year. Selling, general and administrative expenses increased primarily as a result of increases in personnel costs (\$3.0 million), facility costs (\$673,000), depreciation and amortization (\$632,000) and advertising and promotion (\$627,000). Personnel and facility

15

costs increased primarily due to costs related to the businesses we acquired in July 2005, as well as due to increases in personnel costs with respect to the staffing of our new initiatives, including Calvin Klein women's suits and dresses, Sean John women's sportswear and Exsto. Facility costs also increased as a result of the additional space leased in our Secaucus warehouse facility that was added in August 2005. Depreciation and amortization expense increased primarily as a result of the amortization of the identifiable intangibles we acquired. Advertising and promotion increased due to contractual advertising under the licenses that we added as a result of the acquisitions and the new Calvin Klein women's suit license.

Interest and finance charges, net for the three month period ended July 31, 2006 were \$1.3 million compared to \$527,000 for the comparable period last year. Interest expense increased due to the term loan entered into in connection with financing our two acquisitions and less interest income on lower average cash balances than in the comparable period in the prior year.

Income tax benefit for the three months ended July 31, 2006 was \$1.3 million compared to \$77,000 in the comparable period last year. The effective rate for the current period was 42.5% compared to 20.4% for the comparable prior period. We compute our tax expense based on an estimated annual tax rate. At the end of our first fiscal quarter of the prior year, we estimated the annual tax rate to be 43.0%. This estimated rate was decreased in the second quarter to an estimated annual tax rate of 42.0%. As a result, our effective rate for the three months ended July 31, 2005 was 20.4%. The lower effective rate for the three months ended July 31, 2005 reflects the full year's effect of the change in our estimated annual rate.

SIX MONTHS ENDED JULY 31, 2006 COMPARED TO SIX MONTHS ENDED JULY 31, 2005

Net sales for the six months ended July 31, 2006 increased to \$83.5 million from \$68.3 million in the same period last year. Net sales of licensed apparel increased to \$56.0 million from \$36.2 million, primarily as a result of sales of Calvin Klein suits and outerwear (\$12.1 million). Net sales of non-licensed

apparel decreased to \$27.5 million from \$32.1 million, primarily due to a decrease in private label men's outerwear sales (\$6.1 million) offset, in part, by the initial shipments of Exsto private label product (\$1.8 million) which were primarily shipped in July 2006.

Gross profit increased to \$17.5 million, or 21.0% of net sales, for the six month period ended July 31, 2006, from \$13.7 million, or 20.0% of net sales, in the same period last year. The gross profit percentage in our licensed apparel segment was 23.0% in the six month period ended July 31, 2006 compared to 19.9% in the same period last year. Sales of Calvin Klein suits and outerwear resulted in an increase in both margin dollars and our gross profit as a percentage of sales. The gross profit percentage in our non-licensed segment was 16.9% in the six month period ended July 31, 2006 compared to 20.2% in the same period last year. The decrease in the gross profit percentage is primarily attributable to lower margin on the sales of products under the Marvin Richards label than under our other owned labels.

Selling, general and administrative expenses increased \$12.3 million to \$34.0 million in the six month period ended July 31, 2006 from \$21.7 million in the same period last year. Selling, general and administrative expenses increased primarily as a result of increases in personnel costs (\$5.5 million), facility costs (\$1.6 million), depreciation and amortization (\$1.4 million) and advertising and promotion (\$1.3 million). Personnel and facility costs increased primarily due to costs related to the businesses we acquired in July 2005, as well as due to increases in personnel costs with respect to the staffing of our new initiatives, including Calvin Klein women's suits and dresses, Sean John women's sportswear and Exsto. Facility costs also increased as a result of the additional space leased in our Secaucus warehouse facility that was added in August 2005. Depreciation and amortization expense increased as a result of the amortization of the identifiable intangibles we acquired. Advertising and promotion increased primarily due to contractual advertising under the licenses that we added as a result of the acquisitions and the new Calvin Klein women's suit license.

Interest and finance charges, net for the six month period ended July 31, 2006 were \$1.9 million compared to \$530,000 for the comparable period last year. Interest expense increased due to the new term loan entered into in connection with the acquisitions and less interest income on lower average cash balances than in the comparable period in the prior year.

Income tax benefit for the six months ended July 31, 2006 was \$7.8 million compared to \$3.6 million in the comparable period last year. The effective rate for the current period was 42.5% compared to 42.0% for the comparable prior period.

16

LIQUIDITY AND CAPITAL RESOURCES

Our primary cash requirements are to fund our seasonal build up in inventories and accounts receivable, primarily during our second and third fiscal quarters each year. Due to the seasonality of our business, we generally reach our maximum borrowing under our asset-based credit facility during our third fiscal quarter. The primary sources to meet our cash requirements are borrowings under this credit facility and cash generated from operations. At July 31, 2006, we had cash and cash equivalents of \$728,000 and outstanding borrowings of \$67.1 compared to cash and cash equivalents of \$1.2 million and outstanding borrowings of \$77.3 million at July 31, 2005. Our borrowings were lower as of July 31, 2006 as a result of the application of the net proceeds of our private placement to the outstanding balance under our revolving credit line.

Private Placement

On July 13, 2006, we completed a private placement of our common stock and five-year warrants to purchase our common stock pursuant to a securities purchase agreement between us and a group of investors resulting in aggregate proceeds to the Company of \$15,165,000. The net proceeds of this placement were used to repay a portion of our outstanding balance under our revolving credit line.

We issued 1,500,000 shares of our common stock to the investors at a price of \$10.11 per share. We also issued to the investors warrants to purchase an

aggregate of up to 375,000 shares of our common stock, exercisable beginning six months after the closing date of the private placement, at an exercise price of \$11.00 per share, subject to adjustment upon the occurrence of specified events, including customary weighted average price anti-dilution adjustments.

For two years after the closing date of the private placement, the investors will, subject to exceptions and qualifications specified in the purchase agreement, have a right of first refusal with respect to the proposed sale by us of our equity or equity equivalent securities at an effective price per share of \$10.00 or less.

We also entered into a registration rights agreement with the investors, in which we agreed to file a registration statement with the Securities and Exchange Commission to register under the Securities Act of 1933, as amended, resales from time to time of the shares, any warrant shares issued upon exercise of the warrants and an additional 500,000 shares of our common stock sold to the investors by Mr. Aron Goldfarb, the Company's founder and father of our Chief Executive Officer, on July 13, 2006. We filed the registration statement within the required time period and the registration statement has been declared effective by the SEC.

Financing Agreement

We have a financing agreement with The CIT Group/Commercial Services, Inc., as Agent, for a consortium of banks. The financing agreement, which expires on July 11, 2008, is a senior secured credit facility providing for borrowings in the aggregate principal amount of up to \$195.0 million. The facility consists of a revolving line of credit and a term loan.

The revolving line of credit provides for a maximum line ranging from \$45 million to \$165 million at specific times during the year, provided that there are no borrowings outstanding for at least 45 days during the period from December 1 through April 30 each year. We satisfied this requirement for the most recent period. Amounts available under the line are subject to borrowing base formulas and over advances as specified in the financing agreement. Borrowings under the line of credit bear interest at our option at the prime rate or LIBOR plus 2.25%.

The amount borrowed under the line of credit varies based on our seasonal requirements. As of July 31, 2006, direct borrowings were \$41.2 million and our contingent liability under open letters of credit was approximately \$20.8 million compared to direct borrowings of \$46.5 million and contingent liability under open letters of credit of \$26.5 million as of July 31, 2005.

The term loan in the original principal amount of \$30 million is payable over three years with eleven quarterly installments of principal in the amount of \$1,650,000. Payment of quarterly installments began on December 31, 2005, with the remaining balance of \$11,850,000 due on maturity of the loan. Mandatory prepayments are required under the term loan commencing with the fiscal year ending January 31, 2007 to the extent of 50% of excess cash flow, as defined. The term loan bears interest, at our option, at prime plus 1% or LIBOR plus

17

3.25%. As of August 1, 2006, the term loan bore interest at prime plus 1%, or 9.25% per year. The balance due on the term loan at July 31, 2006 was \$25,050,000.

The financing agreement requires us, among other things, to maintain tangible net worth at specified levels, achieve specified earnings before interest, taxes, depreciation and amortization and maintain minimum fixed charge coverage ratios as defined. It also limits capital expenditures and payments for cash dividends and stock redemption to \$1.5 million plus an additional amount for stock redemptions based on the proceeds of sales of equity securities. As of July 31, 2006, we were in compliance with these covenants. The financing agreement is collateralized by all of our assets.

Subsidiary Loan

PT Balihides, our inactive Indonesian subsidiary, had a separate credit facility with an Indonesian bank. In December 2002, we closed the manufacturing facility operated by this subsidiary. The notes payable under this facility represent

borrowings as of July 31, 2006 of approximately \$770,000. The loan is collateralized by the property, plant, and equipment of this subsidiary. No other G-III entity has guaranteed this loan. We continue to be in discussions with the bank regarding settlement of this debt.

Cash from Operating Activities

We used \$55.7 million of cash from operating activities in the six months ended July 31, 2006, primarily as a result of our net loss of \$10.6 million and increases of \$50.8 million in inventory, \$12.6 million in accounts receivable, \$9.7 million in prepaid and refundable income taxes and \$6.0 million in prepaid expenses, offset, in part, by an increase in accounts payable and accrued expenses of \$31.4 million. The increases in these operating cash flow items are consistent with our seasonal pattern. We typically have a net loss through our first two fiscal quarters. During the second quarter, we build inventory for the fall shipping season accounting for the increase in inventory. The fall shipping season begins in the latter part of our second quarter. The increase in our accounts receivable reflects the beginning of the sales shipped for fall. The decrease in income taxes payable is attributable to income taxes paid subsequent to year end as a result of our fiscal 2006 income and the increase in our prepaid and refundable income taxes is a result of the tax benefit recognized on our loss through the six months. The increase in prepaid expenses is primarily a result of prepaid royalties and advertising under our license agreements. The increase in accounts payable and accrued expenses is attributable to the purchasing activity for the fall season.

Cash from Investing Activities

We used \$3.8 million of cash in investing activities in the six months ended July 31, 2006. We paid \$3.3 million to the sellers of the acquired companies as contingent purchase price based on attaining performance goals as defined in the respective purchase agreements. The sellers are entitled to earn-out payments through the year ended January 31, 2009. We had capital expenditures of \$499,000 in the six months ended July 31, 2006.

Cash from Financing Activities

Cash from financing activities provided \$53.2 million in the six months ended July 31, 2006 primarily from net proceeds to the Company of \$15.0 million from our private placement and net amounts borrowed under our credit facility of \$41.2 million. Borrowings under the credit line were used to finance our inventory purchases for the upcoming fall season and for other working capital purposes. During the six months ended July 31, 2006, we repaid \$3.3 million of our term loan which represents two quarterly installment payments.

Proposed Public Offering

On May 8, 2006, we filed a registration statement with the Securities and Exchange Commission for a proposed offering of 4,000,000 shares of our common stock pursuant to which we would have offered 3,000,000 shares and selling stockholders would have offered 1,000,000 shares. The registration statement was withdrawn on July 12, 2006.

18

CRITICAL ACCOUNTING POLICIES

Our discussion of results of operations and financial condition relies on our consolidated financial statements that are prepared based on certain critical accounting policies that require management to make judgments and estimates that are subject to varying degrees of uncertainty. We believe that investors need to be aware of these policies and how they impact our financial statements as a whole, as well as our related discussion and analysis presented herein. While we believe that these accounting policies are based on sound measurement criteria, actual future events can and often do result in outcomes that can be materially different from these estimates or forecasts. The accounting policies and related estimates described in our Annual Report on Form 10-K for the year ended January 31, 2006 are those that depend most heavily on these judgments and estimates. As of July 31, 2006, there have been no material changes to our critical accounting policies.

In June 2006, the Financial Accounting Standards Board issued FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"). FIN 48 clarifies the accounting for uncertainty in income taxes recognized in our financial statements in accordance with FASB Statement No. 109 "Accounting for Income Taxes." FIN 48 also prescribes a recognition threshold and measurement attribute for the financial statement recognition and measurement of a tax position taken or expected to be taken in a return, as well as guidance on derecognition, classification, interest and penalties and financial statement reporting disclosures. FIN 48 is effective for fiscal years beginning after December 15, 2006. We are currently evaluating the requirements and impact of FIN 48 on our consolidated financial statements, and will adopt the provisions on February 1, 2007.

We adopted SFAS 123R on February 1, 2006 using the modified prospective method. Under this method, we are required to recognize compensation cost, on a prospective basis, for the portion of outstanding awards for which the requisite service has not yet been rendered as of February 1, 2006, based upon the grant-date fair value of those awards calculated under SFAS 123 for pro forma disclosure purposes. Under SFAS 123R, we are required to measure the cost of services received in exchange for stock options and similar awards based on the grant-date fair value of the award and recognize this cost in the income statement over the period during which an award recipient is required to provide service in exchange for the award. The pro forma disclosures previously permitted under SFAS 123 no longer will be an alternative to financial statement recognition.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

There are no material changes to the disclosure made with respect to these matters in our Annual Report on Form 10-K for the year ended January 31, 2006.

ITEM 4. CONTROLS AND PROCEDURES

As of the end of the period covered by this report, our management, including the Chief Executive Officer and Chief Financial Officer, carried out an evaluation of the effectiveness of the design and operation of the Company's disclosure controls and procedures (as such term is defined in Rule 13a-15(e) under the Exchange Act). Based on that evaluation, the Company's Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures are designed to ensure that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is (i) recorded, processed, summarized and reported, within the time periods specified in the Commission's rules and forms and (ii) accumulated and communicated to our management, including our principal executive and principal financial officers, as appropriate to allow timely decisions regarding required disclosure. During our last fiscal quarter, there were no changes in our internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

19

PART II

ITEM 1A. RISK FACTORS

In addition to the other information set forth in this report, you should carefully consider the factors discussed in "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended January 31, 2006, which could materially affect our business, financial condition or future results. The risks described in our Annual Report on Form 10-K are not the only risks facing our company. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially adversely affect our business, financial condition and/or operating results.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF STOCKHOLDERS

- (a) Our Annual Meeting of Stockholders was held on June 8, 2006.
- (b) The following matters were voted on and approved by our stockholders at the Annual Meeting:

(i) The election of nine directors to serve for the ensuing year. The following nominees were elected as directors (with our stockholders having voted as set forth below):

NOMINEE	VOTES FOR	WITHHELD AUTHORITY TO VOTE
Morris Goldfarb	10,930,857	8,707
Sammy Aaron	10,867,437	72,127
Thomas J. Brosig	10,213,449	726,115
Pieter Deiters	10,931,157	8,407
Alan Feller	10,890,992	48,572
Carl Katz	10,930,357	9,207
Laura H. Pomerantz	10,866,937	72,627
Willem van Bokhorst	10,930,857	8,707
Richard White	10,930,857	8,707

(ii) The approval of an amendment to our certificate of incorporation that increased our number of authorized shares of common stock from 20,000,000 to 40,000,000:

FOR: 10,565,482
AGAINST: 84,152
ABSTENTIONS: 289,930

(iii) The ratification of the appointment of Ernst & Young LLP as our independent certified public accountants for the fiscal year ending January 31, 2007. Our stockholders voted as follows:

> FOR: 10,936,454 AGAINST: 2,295 ABSTENTIONS: 815

> > 20

ITEM 6. EXHIBITS

- 3.1 Certificate of Amendment of Certificate of Incorporation, dated June 8, 2006
- 10.1 Lease agreement dated June 29, 2006 between The Realty Associates Fund VI, LP and G-III Apparel Group, Ltd.
- 31.1 Certification by Morris Goldfarb, Chief Executive Officer of G-III Apparel Group, Ltd., pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, in connection with G-III Apparel Group, Ltd.'s Quarterly Report on Form 10-Q for the fiscal quarter ended July 31, 2006.
- 31.2 Certification by Neal S. Nackman, Chief Financial Officer of G-III Apparel Group, Ltd., pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, in connection with G-III Apparel Group, Ltd.'s Quarterly Report on Form 10-Q for the fiscal quarter ended July 31, 2006.
- 32.1 Certification by Morris Goldfarb, Chief Executive Officer of G-III Apparel Group, Ltd., pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, in connection with G-III Apparel Group, Ltd.'s Quarterly Report on Form 10-Q for the fiscal quarter ended July 31, 2006.
- 32.2 Certification by Neal S. Nackman, Chief Financial Officer of G-III Apparel Group, Ltd., pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, in connection with G-III Apparel Group, Ltd.'s Quarterly Report on Form 10-Q for the fiscal quarter ended July 31, 2006.

21

registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

G-III APPAREL GROUP, LTD. (Registrant)

Date: September 13, 2006 By: /s/ Morris Goldfarb

Morris Goldfarb

Chief Executive Officer

Date: September 13, 2006 By: /s/ Neal S. Nackman

Neal S. Nackman

Chief Financial Officer

CERTIFICATE OF AMENDMENT

OF

CERTIFICATE OF INCORPORATION

OF

G-III APPAREL GROUP, LTD.

* * * * *

G-III APPAREL GROUP, LTD., a corporation organized and existing under and by virtue of the General Corporation Law of the State of Delaware (hereinafter called the "Corporation"), DOES HEREBY CERTIFY:

FIRST: That the Board of Directors of the Corporation duly adopted a resolution proposing and declaring advisable the following amendment to the Certificate of Incorporation (the "Certificate of Incorporation") of the Corporation:

RESOLVED, that the Certificate of Incorporation be amended pursuant to a Certificate of Amendment of Certificate of Incorporation (the "Certificate of Amendment"), such that Paragraph (A) of Article Fourth of the Certificate of Incorporation be amended to read in its entirety as follows:

FOURTH: A. Authorized Capital Stock. The total number of shares of all classes of stock which this Corporation shall have authority to issue is FORTY-ONE MILLION (41,000,000) shares, consisting of ONE MILLION (1,000,000) shares of Preferred Stock, par value \$.01 per share (hereinafter, the "Preferred Stock"), and FORTY MILLION (40,000,000) shares of Common Stock, par value \$.01 per share (hereinafter, the "Common Stock").

SECOND: The Certificate of Amendment herein certified was duly adopted at the annual meeting of the stockholders of the Corporation duly called and held, upon notice in accordance with Section 222 of the General Corporation Law of the State of Delaware, at which meeting the necessary number of shares as required by statute voted in favor of the amendment to the Certificate of Incorporation.

IN WITNESS WHEREOF, the undersigned has caused this Certificate of Amendment of Certificate of Incorporation to be signed, under penalties of perjury, and the facts stated herein are true and correct.

Dated: June 8, 2006

By: /s/ Morris Goldfarb

Name: Morris Goldfarb

Title: Chief Executive Officer

STANDARD INDUSTRIAL LEASE

THIS INDUSTRIAL LEASE AGREEMENT ("LEASE") is made and entered into as of the 29th day of June, 2006 by and between THE REALTY ASSOCIATES FUND VI, L.P., a Delaware limited partnership ("LANDLORD"), and G-III APPAREL GROUP LTD., a Delaware corporation ("TENANT").

1. BASIC LEASE PROVISIONS.

- 1.1 BUILDING ADDRESS: 308 Herrod Boulevard, South Brunswick, New Jersey ("Building").
 - 1.2 PREMISES ADDRESS: 308 Herrod Boulevard, South Brunswick, New Jersey.
- 1.3 APPROXIMATE RENTABLE SQUARE FEET OF PREMISES: 305,477, as shown on Exhibit A comprised of (i) Unit A consisting of 105,603 square feet ("Unit A"), (ii) Unit B consisting of 99,611 square feet ("Unit B"), and (iii) Unit C consisting of 100,263 square feet ("Unit C") (Unit A, Unit B and Unit C are collectively referred to herein as the "Premises").
- 1.4 APPROXIMATE RENTABLE SQUARE FEET OF BUILDING: 305,477 rentable square feet.
- 1.5 PERMITTED USE: Warehousing, light assembly and distribution of apparel, including pressing, cleaning, embroidery, screen printing, repair of garments, administrative offices and other similar uses related to wholesale apparel operations, and a portion of the Premises for retail sales of warehoused goods if such use is permitted under applicable law, subject to the requirements and limitations contained in Article 4.
- 1.6 TERM: Seven (7) years from the Unit C Commencement Date (as that term is hereinafter defined) unless Tenant exercises its Option to Terminate the Lease with respect to Unit C in accordance with Paragraph 10 of the Addendum, in which event it shall be seven (7) years from the Unit A Commencement Date (as that term is hereinafter defined).
- 1.7 COMMENCEMENT DATES: October 1, 2006 with respect to Unit B (the "Unit B Commencement Date"); January 1, 2007 with respect to Unit A (the "Unit A Commencement Date"); and February 1, 2007 with respect to Unit C (the "Unit C Commencement Date").
- 1.8 MONTHLY BASE RENT: Commencing on the Unit B Commencement Date (subject to the abatement provided in Paragraph 4 of the Addendum), Monthly Base Rent for Unit B shall be \$33,203.67. Commencing on the Unit A Commencement Date (subject to the abatement provided in Paragraph 4 of the Addendum), in addition to Monthly Base Rent for Unit B, Monthly Base Rent for Unit A shall be \$35,201.00. Commencing on the Unit C Commencement Date (subject to the abatement provided in Paragraph 4 of the Addendum), in addition to Monthly Base Rent for Unit B and Unit A, Monthly Base Rent for Unit C shall be \$33,421.00.
- 1.9 BASE RENT PAID UPON EXECUTION: \$68,404.67 for the first full month of the Term of this Lease for which Base Rent is due for Units A and B. Furthermore, if Tenant does not elect to terminate the Lease with respect to Unit C on or before October 31, 2006 in accordance with Paragraph 10 of the Addendum, then Tenant shall deposit with Landlord, on or before November 10, 2006, \$33,421.00 for the first full month of the Term of this Lease for which Base Rent is due for Unit C.
 - 1.10 TENANT'S PERCENTAGE SHARE: 100%. See Addendum Paragraph 6
- 1.11 SECURITY DEPOSIT: \$136,809.34 (provided, if Tenant does not elect to terminate the Lease with respect to Unit C on or before October 31, 2006 in accordance with Paragraph 10 of the Addendum, then Tenant shall deposit with Landlord, on or before November 10, 2006, an additional security deposit in the amount of \$66,842.00). See Addendum Paragraph 7.

1.12 INTENTIONALLY DELETED

- 1.13 REAL ESTATE BROKERS: Kwartler Associates, Inc. and CB Richard Ellis, Inc. representing Landlord and Team Resources, Inc. representing Tenant.
 - 1.14 ADDRESSES FOR NOTICES:

LANDLORD: The Realty Associates Fund VI, L.P.

c/o Kwartler Associates

2 North Street

Waldwick, New Jersey 07463

WITH A COPY TO: TA Associates Realty

28 State Street

Boston, Massachusetts 02109 Attention: Asset Manager

TENANT: G-III Apparel Group Ltd.

512 Seventh Avenue New York, New York 10018 Attention: Wayne Miller, COO

1.15 ATTACHMENTS: Addendum; Exhibit A (Outline of Premises), Exhibit B (Verification Letter), Exhibit C (Rules and Regulations), Exhibit D (Letter of Credit), Exhibit E (License for Satellite Dish); Exhibit F (Landlord's Waiver and Consent), Exhibit G (Coverall Construction LLC Letter dated June 27, 2006), Schedule 1 (Tenant's Initial Work).

2. PREMISES. The Premises are hereby leased to Tenant to have and to hold for the Term of this Lease, subject to the terms, covenants and conditions of this Lease. Tenant accepts the Premises in its condition as of the respective Commencement Dates, subject to all applicable laws, ordinances, regulations, covenants, conditions, restrictions and easements, and, except as may be otherwise expressly provided herein, Landlord shall not be obligated to make any repairs or alterations to the Premises. The Premises is depicted on Exhibit A attached hereto. The "PROJECT" consists of the Premises, the Building and the land upon which the same are located.

SEE ADDENDUM PARAGRAPHS 1 AND 2

3. TERM.

- 3.1 TERM AND COMMENCEMENT DATE. The Term and Commencement Dates of this Lease are as specified in Sections 1.6 and 1.7. The Commencement Dates set forth in Section 1.7 are estimated Commencement Dates. The actual Commencement Dates shall be the date possession of each of the respective Units of the Premises is tendered to Tenant; provided, however, that if the Unit C Commencement Date (or the Unit A Commencement Date if Tenant exercises its option to terminate the Lease with respect to Unit C) is other than the first day of a calendar month, the Term of this Lease shall be computed from the first day of the calendar month following the Unit C Commencement Date (or the Unit A Commencement Date, if applicable). When the actual Commencement Dates are established by Landlord, Tenant shall, within ten (10) days after Landlord's request, complete and execute the letter attached hereto as Exhibit B and deliver it to Landlord.
- 3.2 DELAY IN POSSESSION. Notwithstanding the estimated Commencement Dates specified in Section 1.7, if for any reason Landlord cannot deliver possession of any of the Units of the Premises to Tenant on said dates, Landlord shall not be subject to any liability therefor, nor shall such failure affect the validity of this Lease or the obligations of Tenant hereunder; provided, however, in such a case, Tenant shall not be obligated to pay rent or perform any other obligation of Tenant under this Lease, except as may be otherwise provided in this Lease, until possession of the applicable portion of the Premises is tendered to Tenant. If Landlord shall not have tendered possession of any of the Units of the Premises to Tenant within thirty (30) days following the estimated Commencement Date for said Unit as set forth in Section 1.7 (hereinafter, the "Delay Credit Date"), as the same may be adjusted for any Force Majeure Events (defined hereinbelow) and/or any delays caused by acts or omissions of Tenant, Tenant's agents, employees and contractors of which Tenant has received at least three (3) days' prior written notice from Landlord (hereinafter, a "Tenant Delay"), then Tenant shall be entitled to an abatement of Base Rent with respect to the applicable Premises (the "Delay Credit") in an amount equal to one-thirtieth (1/30th) of the Base Rent due for the first full calendar month of the Term with respect to the applicable Premises (not reduced by any abated rent, conditionally waived rent, free rent or similar rental concessions, if any) for each day following the Delay Credit Date until such time as the applicable Premises is delivered to Tenant. The Delay Credit, if any, shall be applied to the next installment(s) of Base Rent or additional rent coming due

under the Lease following the application of any abatement of Base Rent as provided in Paragraph 4 of the Addendum. Furthermore, if Landlord shall not have tendered possession of all of the Premises to Tenant on or before March 1, 2007 (the "Outside Commencement Date"), as the same may be extended for any Force Majuere Events and Tenant Delay, Tenant may, at Tenant's option, by notice in writing to Landlord at any time after the Outside Commencement Date, terminate the term of this Lease so long as Tenant's written notice is delivered to Landlord prior to Landlord's delivery of the Premises to Tenant. If Tenant terminates the term of this Lease as provided in the preceding sentence, the parties shall be discharged from all obligations hereunder, except that Landlord shall return any money previously deposited with or paid to Landlord by Tenant. If Landlord is unable to deliver possession of any of the Premises to Tenant on the applicable Commencement Date due to a "Force Majeure Event," the applicable Commencement Date shall be extended by the period of the delay caused by the Force Majeure Event.

4. USE.

4.1 PERMITTED USE. The Premises shall be used for the Permitted Use only and for no other purpose. Tenant shall not (a) permit any animals or pets to be brought to or kept in the Premises, (b) install any antenna, dish or other device on the roof of the Building or outside of the Premises, except as expressly provided in Paragraph 3 of the Addendum, (c) make any penetrations into the roof of the

2

Building, (d) place loads upon floors, walls or ceilings in excess of the load such items were designed to carry, (e) place or store, nor permit any other person or entity to place or store, any property, equipment, materials, supplies or other items outside of the Building in which the Premises is located without the prior written consent of Landlord, such consent not to be unreasonably withheld, conditioned or delayed, or (f) change the exterior of the Premises or the Building in which the Premises is located.

SEE ADDENDUM PARAGRAPH 3

4.2 COMPLIANCE WITH LAWS. Except with respect to the Landlord's Work described in Paragraph 1 of the Addendum and as expressly set forth in this Section 4.2, Tenant shall, at Tenant's sole expense, promptly comply with all applicable laws, ordinances, rules, regulations, orders, certificates of occupancy, conditional use or other permits, variances, covenants, conditions, restrictions, easements, and requirements of any fire insurance underwriters, rating bureaus or government agencies, now in effect or which may hereafter come into effect (collectively "REGULATIONS"), whether or not they reflect a change in policy from that now existing, during the Term or any part of the Term hereof, relating in any manner to the Premises or the occupation and use by Tenant of the Premises. Notwithstanding anything to the contrary set forth above, Tenant shall not be required to comply with any such Regulations to the extent that such Regulations require repairs or alterations which constitute Landlord Repairs as defined in Article 11 hereof. Landlord, at its sole cost and expense, shall comply with all Regulations which require repairs or alterations which constitute Landlord Repairs unless such repairs or alterations are required by Tenant's specific use of the Premises or by any alterations or improvements to the Premises made by Tenant or on Tenant's behalf. In addition, and notwithstanding anything to the contrary set forth above, to the extent any Regulation requires that any repair or alteration be made to the Premises which constitutes a Capital Improvement, as hereinafter defined, unless such repair or alteration is required by Tenant's specific use of the Premises or by any alterations or improvements to the Premises made by Tenant or on Tenant's behalf, Landlord shall comply with such Regulation and the cost of such Capital Improvement which is necessary to comply with such Regulation shall be amortized over the useful life of such Capital Improvement as determined in accordance with Generally Accepted Accounting Principles ("GAAP"), together with an interest factor on the unamortized cost of such item equal to ten percent (10%) per annum and Tenant shall be responsible for the amortized portion of the cost of such Capital Improvement applicable to the remainder of the Term, payable monthly. Except with respect to the Landlord's Work described in Paragraph 1 of the Addendum and as expressly set forth in this Section 4.2, Tenant shall, at Tenant's sole expense, comply with all requirements of the Americans With Disabilities Act ("ADA") that relate to the Premises, and all federal, state and local laws and regulations governing occupational safety and health.

Notwithstanding the foregoing, in the event any applicable authority requires the Premises to be brought into compliance with ADA standards which were in effect but not enforced prior to the Commencement Date for any reason other than Tenant's specific use of the Premises (if same is other than for basic warehouse uses) or by any alterations or improvements to the Premises made by Tenant or on Tenant's behalf, then Landlord shall be responsible for all costs associated therewith

- 4.3 NUISANCE; PERMITS. Tenant shall not permit any objectionable or unpleasant odors, smoke, dust, gas, noise or vibrations to emanate from the Premises, or take any other action that would constitute a nuisance or would disturb, unreasonably interfere with or endanger Landlord or any other tenants of the Project. Tenant shall obtain, at its sole expense, any permit or other governmental authorization required to operate its business from the Premises. Landlord shall not be liable for the failure of any other tenant or person to abide by the requirements of this Section, and Tenant shall not be excused from the performance of its obligations under this Lease due to such a failure.
- 4.4 COMPLIANCE WITH LAWS ON COMMENCEMENT. The Premises (except with respect to any alterations or improvements to be made by Tenant but including the Landlord's Work, more particularly described in Paragraph 1 of the Addendum) shall be in compliance with applicable Regulations in effect on the date on which Landlord tenders possession of the Premises to Tenant. To the extent that the Project (in the condition existing on the date on which Landlord tenders possession of the Premises to Tenant but without regard to any alterations or improvements to be made by Tenant or Tenant's use of the Premises) is in violation of any Regulation on the date on which Landlord tenders possession of the Premises to Tenant and Landlord is obligated pursuant to a final determination to undertake action in order to comply with such Regulation, then in such event Landlord shall be obligated to undertake such remedial action in order to comply with such Regulation, at Landlord's sole cost and expense.
- 5. BASE RENT. Tenant shall pay Base Rent in the amount set forth in Section 1.8. The first month's Base Rent, the Security Deposit, and the first monthly installment of estimated Operating Expenses (as hereinafter defined), if applicable, shall be due and payable on the date this Lease is executed by Tenant, and Tenant promises to pay to Landlord in advance, without notice or demand, deduction, abatement or set-off (unless expressly set forth in this Lease), monthly installments of Base Rent with respect to each Unit of the Premises on or before the first day of each calendar month succeeding the applicable Commencement Date. Base Rent for any partial month during the Term shall be prorated. All payments required to be made by Tenant to Landlord hereunder shall be payable at such address as Landlord may specify from time to time by written notice delivered in accordance herewith.

SEE ADDENDUM PARAGRAPHS 4 AND 5

3

6. OPERATING EXPENSE PAYMENTS.

6.1 OPERATING EXPENSES. Except for Landlord Repairs, as more particularly set forth in Article 11, Tenant shall pay Tenant's Percentage Share of the Operating Expenses for the Project. For the purposes of this Lease, the term "OPERATING EXPENSES" shall mean all expenses and disbursements of every kind (subject to the limitations set forth below) which Landlord incurs, pays or becomes obligated to pay in connection with the ownership, operation, and maintenance of the Project, including, but not limited to, the following: (a) wages and salaries (including management fees) of all employees, agents, consultants and other individuals or entities engaged in the operation, repair, replacement, maintenance, and security of the Project, including taxes, insurance and benefits relating thereto; (b) all supplies and materials used in the operation, maintenance, repair, replacement, and security of the Project; (c) annual cost of all Capital Improvements (as defined below) made to the Project which although capital in nature can reasonably be expected to reduce the normal operating costs of the Project, as well as all Capital Improvements made in order to comply with any law now or hereafter promulgated by any governmental authority, as amortized over the useful economic life of such improvements as determined by Landlord in its reasonable discretion (without regard to the period over which such improvements may be depreciated or amortized for federal income tax purposes); (d) cost of all utilities paid by Landlord; (e) cost of repairs, replacements and general maintenance of the

Project (including all truck court areas, paving and parking areas, lighting facilities, fences, gates, water lines, sewer lines, rail spur areas and any other item Landlord is obligated to repair or maintain), other than costs necessary to assure the structural soundness of the roof, foundation and exterior walls of the Project which are payable solely by Landlord under Article 11; (f) cost of service or maintenance contracts with independent contractors for the operation, maintenance, repair, replacement or security of the Project (including, without limitation, alarm service, exterior painting, trash collection, snow, ice, debris and waste removal and landscape maintenance); (g) the cost of all accounting fees, management fees, legal fees and consulting fees attributable to the operation, ownership, management, maintenance or repair of the Project; (h) payments made by Landlord under any easement, license, operating agreement, declaration, restrictive covenant or other agreement relating to the sharing of costs among property owners; (i) the cost of all business licenses, permits or similar fees relating to the operation, ownership, repair or maintenance of the Project; and (j) the cost of any other item the cost of which is stated in this Lease to be an Operating Expense. For purposes of this Lease, a "CAPITAL IMPROVEMENT" shall be an improvement to the Project that Landlord is obligated to make pursuant to this Lease, the cost of which is not fully deductible in the year incurred in accordance with GAAP; provided, however, that, at Landlord's option, the cost of painting all or part of the Project and resurfacing and restriping roadways and parking areas shall be treated as an expense and not as a Capital Improvement. Real Property Taxes (as defined below) and Insurance Costs (as defined below) shall be reimbursed to Landlord as provided below and shall not be treated as an Operating Expense.

SEE ADDENDUM PARAGRAPH 6

6.2 OPERATING EXPENSE EXCLUSIONS. Notwithstanding anything to the contrary contained herein, Operating Expenses shall not include the following: (i) legal and auditing fees (other than those fees reasonably incurred in connection with the maintenance and operation of all or any portion of the Project), leasing commissions, advertising expenses and similar costs incurred in connection with the leasing of the Project; (ii) depreciation of the Building or any other improvements situated within the Project; (iii) any items for which Landlord is actually reimbursed by insurance; (iv) costs of repairs or other work necessitated by fire, windstorm or other casualty (excluding any deductibles) and/or costs of repair or other work necessitated by the exercise of the right of eminent domain to the extent insurance proceeds or a condemnation award, as applicable, is actually received by Landlord for such purposes; (v) costs associated with the investigation and/or remediation of Hazardous Materials (hereafter defined) present in, on or about any portion of the Premises, unless such costs and expenses are the responsibility of Tenant as provided in Article 27 hereof, in which event such costs and expenses shall be paid solely by Tenant in accordance with the provisions of Article 27 hereof; (vi) the cost of Capital Improvements unless made with the reasonable expectation of reducing Operating Expenses or in order to comply with any law first taking effect or interpreted differently on or after the Commencement Date, other than any interest charges for Capital Improvements referred to in Section 6.1(c) hereinabove, any interest or payments on any financing for the Building or the Project and interest and penalties incurred as a result of Landlord's late payment of any invoice; (vii) overhead and profit increment paid to Landlord or to subsidiaries or affiliates of Landlord for goods and/or services in the Project to the extent the same exceeds the costs of such by unaffiliated third parties on a competitive basis; (viii) any payments under a ground lease or master lease; and (ix) except as provided in Section 6.1 above, the cost of Capital Improvements.

6.3 PAYMENT. Tenant shall pay to Landlord Tenant's Percentage Share of Operating Expenses in monthly installments on the first day of each calendar month together with Tenant's payment of Base Rent, in advance, in an amount reasonably estimated by Landlord from time to time. Landlord shall use its best efforts to deliver to Tenant within one hundred eighty (180) days after the expiration of each calendar year a reasonably detailed statement (the "STATEMENT") showing Tenant's Percentage Share of the actual Operating Expenses incurred during such year. Landlord's failure to deliver the Statement to Tenant within said period shall not constitute Landlord's waiver of its right to collect said amounts or otherwise prejudice Landlord's rights hereunder. If Tenant's payments under this Section during said calendar year exceed Tenant's Percentage Share of the actual Operating Expenses incurred as indicated on the Statement, Tenant shall be entitled to credit the amount of such overpayment against Tenant's Percentage Share of Operating Expenses next falling due. If Tenant's payments under this

Section during said calendar year were less than Tenant's Percentage Share as indicated on the Statement, Tenant shall pay to Landlord the amount of the deficiency within thirty (30) days after delivery by Landlord to Tenant of the Statement. Landlord and Tenant shall forthwith adjust between them by cash payment any balance determined to exist with respect to that portion of the last calendar year for which Tenant is responsible for Operating Expenses, notwithstanding that the Term may have terminated before the end of such calendar year; and this provision shall survive the expiration or earlier termination of the Lease.

6.4 AUDITS. If Tenant disputes the amount set forth in the Statement, Tenant shall have the right, at Tenant's sole expense, not later than ninety (90) days following receipt of such Statement, to cause Landlord's books and records with respect to the calendar year which is the subject of the Statement to be audited by a certified public accountant mutually acceptable to Landlord and Tenant. The audit shall take place at the offices of Landlord where its books and records are located at a mutually convenient time during Landlord's regular business hours. Operating Expenses shall be appropriately adjusted based upon the results of such audit, and the results of such audit shall be final and binding upon Landlord and Tenant. Tenant shall have no right to conduct an audit or to give Landlord notice that it desires to conduct an audit at any time Tenant is in default under the Lease. The accountant conducting the audit shall be compensated on an hourly basis and shall not be compensated based upon a percentage of overcharges it discovers. No subtenant shall have any right to conduct an audit, and no assignee shall conduct an audit for any period during which such assignee was not in possession of the Premises. Tenant's right to undertake an audit with respect to any calendar year shall expire ninety (90) days after Tenant's receipt of the Statement for such calendar year, and such Statement shall be final and binding upon Tenant and shall, as between the parties, be conclusively deemed correct, at the end of such ninety (90) day period, unless prior thereto Tenant shall have given Landlord written notice of its intention to audit Operating Expenses for the calendar year which is the subject of the Statement. If Tenant gives Landlord notice of its intention to audit Operating Expenses, it must commence such audit within ninety (90) days after such notice is delivered to Landlord, and the audit must be completed within one hundred twenty (120) days after such notice is delivered to Landlord. If Tenant does not commence and complete the audit within such periods, the Statement which Tenant elected to audit shall be deemed final and binding upon Tenant and shall, as between the parties, be conclusively deemed correct. Tenant agrees that the results of any Operating Expense audit shall be kept strictly confidential by Tenant and shall not be disclosed to any other person or entity.

6.5 OPERATING EXPENSE INCREASES. Notwithstanding anything to the contrary contained in this Article 6, Landlord hereby agrees that for the period commencing October 1, 2006 through December 31, 2007 (the "Fixed Controllable Operating Expense Period"), Operating Expenses (other than utilities, snow removal and management fees; hereinafter, for purposes of this Section 6.5, all Operating Expenses, other than utilities, snow removal and management fees, shall be known collectively as "Controllable Operating Expenses"), shall be deemed to equal \$.20 per square foot of the Premises and such amount shall not be subject to adjustment as generally provided in this Article 6. Furthermore, following the expiration of the Fixed Controllable Operating Expense Period, for each calendar year thereafter during the Term hereof, Tenant shall not be obligated to pay to Landlord its share of Controllable Operating Expenses to the extent that such Controllable Operating Expenses exceed six and one-half percent (6.5%) of the previous calendar year's Controllable Operating Expenses (the "Controllable Operating Expense Cap"); provided, however, that the Controllable Operating Expense Cap shall not apply to the payment by Tenant of (i) utility costs and snow removal costs for the Project (which shall be paid as otherwise provided in this Article 6), and (ii) Project management fees (which shall be paid as set forth hereinbelow). The Controllable Operating Expense Cap shall be applied separately during each year of the Term and shall function as follows: If Controllable Operating Expenses have increased by more that six and one-half percent (6.5%) in a previous calendar year and by less than six and one-half percent (6.5%) in a subsequent calendar year, Landlord shall have the right to pass through to Tenant the increase in Controllable Operating Expenses in such subsequent calendar year that Landlord was unable to pass through in such previous calendar year provided that Landlord does not exceed in any year a maximum increase of six and one-half percent (6.5%) in any calendar year. Management fees for the Project shall be in an amount equal to three percent (3%) of gross revenues annually. For purposes of this Section 6.5, the term "gross revenues" shall mean base rent and all additional rent received by

Landlord for any space in the Project.

- 6.6 ALLOCATION OF EXPENSES. Notwithstanding anything to the contrary in this Article 6, if Tenant leases 100% of the Building, then the definition of Operating Expenses in Section 6.1 above shall be revised to include only management fees in an amount equal to Two Thousand and 00/100 Dollars (\$2,000.00) per month. Real Property Taxes (as defined below) and Insurance Costs (as defined below) shall be reimbursed to Landlord as provided below and shall not be treated as an Operating Expense.
- 7. SECURITY DEPOSIT. The Security Deposit shall be delivered to Landlord upon the execution of this Lease by Tenant and held by Landlord without liability for interest as security for the performance of Tenant's obligations. Landlord may use all or any portion of the Security Deposit for the payment of any past due Base Rent or other charge due hereunder, to cure any default by Tenant or to compensate Landlord for any loss or damage which Landlord may actually suffer thereby. Tenant shall within ten (10) days after written demand therefor deposit cash with Landlord in an amount sufficient to restore the Security Deposit to its full amount. Landlord shall not be required to keep the Security Deposit separate from its general accounts. If Tenant performs Tenant's obligations hereunder, said deposit, or so much thereof as has not heretofore been applied by Landlord, shall be returned to Tenant (or, at Landlord's option, to the last assignee, if any, of Tenant's interest hereunder) at the expiration of the Term hereof,

5

and after Tenant has vacated the Premises. No trust relationship is created herein between Landlord and Tenant with respect to the Security Deposit. The Security Deposit is not an advance payment of rent or a measure of Landlord's damages in the event of Tenant's default. Tenant hereby waives the provisions of any law that is inconsistent with this Section.

SEE ADDENDUM PARAGRAPH 7

8. UTILITIES.

- 8.1 PAYMENT. Tenant shall pay for all water, gas, electricity, telephone, sewer, sprinkler services, refuse and trash collection and other utilities and services used on the Premises, together with any taxes, penalties, surcharges or the like pertaining thereto. Tenant shall contract directly with the applicable public utility for such services.
- 8.2 INTERRUPTIONS. Tenant agrees that Landlord shall not be liable to Tenant for its failure to furnish water, gas, electricity, telephone, sewer, refuse and trash collection or any other utility services or building services when such failure is occasioned, in whole or in part, by repairs, replacements or improvements, by any strike, lockout or other labor trouble, by any accident, casualty or event arising from any cause whatsoever, including the negligence of Landlord, its employees, agents and contractors. Notwithstanding anything contained herein to the contrary, if any interruption of utilities or services caused by Landlord's gross negligence or willful misconduct shall continue for more than five (5) consecutive business days and shall render all or any portion of the Premises unusable for the normal conduct of Tenant's business, and if Tenant does not in fact use or occupy such portion of the Premises, then all Base Rent and additional rent payable hereunder with respect to such portion of the Premises which Tenant does not occupy shall be abated from and after the sixth (6th) consecutive business day until full use of such portion of the Premises is restored to Tenant. Landlord hereby represents and warrants to Tenant that the Premises is currently served by water, electric, telephone and sewer service lines.

9. REAL AND PERSONAL PROPERTY TAXES.

9.1 PAYMENT OF TAXES. Tenant shall pay to Landlord during the Term of this Lease, in addition to Base Rent, Tenant's Percentage Share of all Real Property Taxes (as defined below) assessed against the Project. Real Property Taxes shall be payable by Tenant at the same time, in the same manner and under the same terms and conditions as Tenant pays Operating Expenses. As used herein, the term "REAL PROPERTY TAXES" shall include any form of real estate tax or assessment, general, special, ordinary or extraordinary, improvement bond or bonds imposed on the Project or any portion thereof by any authority having the direct or

indirect power to tax, including any city, county, state or federal government, or any school, agricultural, sanitary, fire, street, drainage or other improvement district thereof, as against any legal or equitable interest of Landlord in the Project or in any portion thereof. Real Property Taxes shall not include income, inheritance and gift taxes.

9.2 PERSONAL PROPERTY TAXES. Tenant shall pay prior to delinquency all taxes assessed against and levied upon trade fixtures, furnishings, equipment and all other personal property of Tenant contained in the Premises or related to Tenant's use of the Premises. If any of Tenant's personal property shall be assessed with Landlord's real or personal property, Tenant shall pay to Landlord the taxes attributable to Tenant within ten (10) days after receipt of a written statement from Landlord setting forth a reasonable allocation of the taxes applicable to Tenant's property.

10. INSURANCE.

10.1 INSURANCE-TENANT.

- (a) Tenant shall obtain and keep in force during the Term of this Lease a commercial general liability policy of insurance with coverages reasonably acceptable to Landlord, which, by way of example and not limitation, protects Tenant and Landlord (as an additional insured) against claims for bodily injury, personal injury and property damage based upon, involving or arising out of the use, occupancy or maintenance of the Premises and Project areas appurtenant thereto. Such insurance shall be on an occurrence basis providing single-limit coverage in an amount not less than \$2,000,000 per occurrence which limit can be satisfied in conjunction with an Umbrella or Excess Liability policy or policies. The policy shall include coverage for liability assumed under this Lease as an "insured contract" for the performance of Tenant's indemnity obligations under this Lease.
- (b) Tenant shall obtain and keep in force during the Term of this Lease "Causes of Loss Special Form" coverage including flood coverage, sprinkler leak coverage, and such endorsements and supplemental coverages as Landlord may reasonably require from time to time. Said insurance shall be written on a one hundred percent (100%) replacement cost basis on Tenant's personal property, all tenant improvements installed at the Premises by Tenant, Tenant's trade fixtures and other Tenant property.
- (c) Tenant shall, at all times during the Term hereof, maintain in effect workers' compensation insurance as required by applicable law.

6

10.2 INSURANCE-LANDLORD.

- (a) Landlord shall obtain and keep in force a policy of general liability insurance with coverage against such risks and in such amounts as are customarily maintained by South Brunswick, New Jersey landlords of facilities similar to the Project ("Local Standards"). Landlord shall also obtain and keep in force during the Term of this Lease a policy or policies of insurance covering loss or damage to the Project in the amount of not less than the full replacement cost thereof (less foundations and footings). The terms and conditions of said policies and the perils and risks covered thereby shall meet Local Standards. In addition, at Landlord's option, Landlord shall obtain and keep in force, during the Term of this Lease, a policy of rental interruption insurance, with loss payable to Landlord, which insurance shall, at Landlord's option, also cover all Operating Expenses and Real Property Taxes. Tenant will not be named as an additional insured in any insurance policies carried by Landlord and shall have no right to any proceeds therefrom. Tenant shall pay, at Tenant's sole expense, any increase in the property insurance premiums for the Project over what was payable immediately prior to the increase to the extent the increase is specified by Landlord's insurance carrier as being caused by the nature of Tenant's use of the Premises or any negligent or willful act or omission of Tenant.
- (b) Tenant shall pay to Landlord during the Term of this Lease, in addition to Base Rent, Operating Expenses and Real Property Taxes, Tenant's Percentage Share of all Insurance Costs (as defined below) paid by Landlord. Insurance Costs shall be payable by Tenant at the same time, in the same manner and under the same terms and conditions as Tenant pays Operating Expenses. For

purposes of this Lease, "INSURANCE COSTS" shall mean the cost of all insurance purchased by Landlord pursuant to section (a) above.

- 10.3 INSURANCE POLICIES. Tenant shall deliver to Landlord certificates of the insurance policies required under Section 10.1 within fifteen (15) days prior to the first Commencement Date of this Lease. Tenant's insurance policies shall not be cancelable or subject to reduction of coverage or other modification except after thirty (30) days prior written notice to Landlord. Tenant shall, at least ten (10) days prior to the expiration of such policies, furnish Landlord with certificates of renewal thereof. Tenant's insurance policies shall be issued by insurance companies authorized to do business in the state in which the Project is located, and said companies shall maintain during the policy term a "General Policyholder's Rating" of at least A and a financial rating of at least "Class X" as set forth in the most recent edition of "Best Insurance Reports." All insurance obtained by Tenant shall be primary to and not contributory with any similar insurance carried by Landlord, whose insurance shall be considered excess insurance only. Landlord and, at Landlord's option, the holder of any mortgage or deed of trust encumbering the Project and any person or entity managing the Project on behalf of Landlord, shall be named as an additional insured on all insurance policies Tenant is obligated to obtain by Section 10.1 above. Tenant shall have the option, either alone or in conjunction with any subsidiaries or affiliates of Tenant, to maintain self insurance and/or provide or maintain any insurance required by this Section under blanket insurance policies maintained by Tenant or provide or maintain insurance through such alternative risk management programs as Tenant may participate in from time to time (such types of insurance programs being herein collectively and severally referred to as "self insurance"), provided the same does not thereby decrease the insurance coverage or limits set forth in this Section or other Tenant insurance requirements set forth elsewhere in the Lease. Any self insurance shall be deemed to contain all of the terms and conditions applicable to such insurance as required in this Article 10, including, without limitation, a full waiver of subrogation as is required in Section 10.4. If Tenant elects to self insure, then with respect to any claims which may result from incidents occurring during the Term of this Lease, such self insurance obligations shall survive the expiration or earlier termination of this Lease to the same extent as the insurance required would survive. Tenant shall be permitted to maintain deductibles in accordance with its standard corporate policies, it being acknowledged and agreed to by Tenant, however, that all such deductible amounts shall be deemed to be self-insurance, and shall be subject to all of the provisions of this Article 10, including, without limitation, a full waiver of subrogation pursuant to Section 10.4. If any of the insurance coverage which Tenant is obligated to carry pursuant to this Lease is under a blanket insurance policy, then such blanket insurance policy shall expressly afford coverage for the Premises and Landlord as required hereunder. Landlord makes no representation to Tenant that the limits or forms of coverage specified above or approved by Landlord are adequate to insure Tenant's property or Tenant's obligations under this Lease, and the limits of any insurance carried by Tenant shall not limit Tenant's obligations or liability under any indemnity provision included in this Lease or under any other provision of this Lease.
- 10.4 WAIVER OF SUBROGATION. Landlord and Tenant hereby waive and shall cause their respective insurance carriers to waive any and all rights of recovery, claims, actions or causes of action against the other for any loss or damage with respect to Tenant's Property, as hereinafter defined, the Building, the Premises, or any contents thereof, including rights, claims, actions and causes of action based on negligence, which loss or damage is (or would have been, had the insurance required by this Lease been carried) covered by insurance. For the purposes of this waiver, any deductible with respect to a party's insurance shall be deemed covered by and recoverable by such party under valid and collectable policies of insurance.
- 11. LANDLORD'S REPAIRS. Landlord shall maintain, at Landlord's sole cost and expense, only the roof of the Building (including the roof membrane), the structural soundness of the foundation of the Building, the structural elements of the elevator serving the Premises and the elevator's motor (subject to Tenant's obligations with respect to the elevator in Section 12.2 below) and the structural elements of the exterior

necessitated by Tenant's misuse, negligence, alterations to the Premises or any breach of its obligations under this Lease. By way of example, and not limitation, the term "exterior walls" as used in this Section shall not include windows, glass or plate glass, doors or overhead doors, special store fronts, dock bumpers, dock plates or levelers, or office entries. Tenant shall, as soon as reasonably possible, give Landlord written notice of any repair required by Landlord pursuant to this Section, after which Landlord shall have a reasonable time in which to complete the repair. Nothing contained in this Section shall be construed to obligate Landlord to seal or otherwise maintain the surface of any foundation, floor or slab, and Tenant shall be obligated to perform maintenance to the warehouse floor necessitated by ordinary wear and tear. Except as herein expressly provided, Tenant expressly waives the benefits of any statute now or hereafter in effect which would otherwise afford Tenant the right to make repairs at Landlord's expense or to terminate this Lease because of Landlord's failure to keep the Premises in good order, condition and repair. Notwithstanding anything to the contrary contained in the Lease, in the event Landlord fails to undertake and diligently pursue any repair to the Premises required to be made by Landlord pursuant to this Article 11 or Section 12.4 below within thirty (30) days following Landlord's receipt of Tenant's written notice, then Tenant shall have the right to make any such repair, provided Landlord has failed to commence and diligently prosecute such repair during such thirty (30) day period. Tenant shall be permitted to make any of the repairs required to be made by Landlord hereunder on an emergency basis without notice to Landlord if such condition requiring repair poses an imminent danger to life or property so long as Tenant attempts by all reasonable means to notify Landlord prior to commencing such repair and notifies Landlord in writing within two (2) business days following the date on which Tenant commenced to make any such repair. Landlord shall reimburse Tenant for the reasonable cost of making any such repair in accordance with this provision within thirty (30) days following Landlord's receipt of an invoice therefor. Tenant shall have no right to set-off any sums incurred by Tenant pursuant to this provision nor shall Tenant be entitled to any abatement of rent except as specifically provided otherwise in this Lease

12. TENANT'S REPAIRS.

12.1 OBLIGATIONS OF TENANT. Subject to Section 12.4 below, Tenant shall, at its sole cost and expense, keep and maintain all parts of the Premises (except those defined as Landlord's Repairs in Article 11 above) in good and sanitary condition, promptly making all necessary repairs and replacements, including but not limited to, windows, glass and plate glass, doors, any special store front or office entry, walls and finish work, elevators, floors and floor coverings, heating and air conditioning systems, dock boards, bumpers, plates, seals, levelers and lights, plumbing work and fixtures (including periodic backflow testing), electrical systems, lighting facilities and bulbs, sprinkler systems, alarm systems, fire detection systems, termite and pest extermination, parking lots, asphalt, curbs, gutters, fencing, landscaped areas, irrigation systems, tenant signage and regular removal of trash and debris, except those repairs or replacements that are required as a direct result of the negligence or willful misconduct of Landlord or Landlord's agents, representatives or contractors. Tenant shall notify Landlord in writing prior to making any repair or performing any maintenance pursuant to this Section, and Landlord shall have the right to designate the contractor Tenant shall use to make any repair or to perform any maintenance on the heating, ventilation and air conditioning systems ("HVAC"), plumbing systems, electrical systems, sprinkler systems, fire alarm systems or fire detection systems located at the Premises providing the rates charged by such contractors are competitive. If Tenant fails to keep the Premises in good condition and repair, Landlord may, upon thirty (30) days prior written notice to Tenant, make any necessary repairs (but shall not be obligated to do so) provided Tenant has failed to commence and diligently prosecute such repair during such thirty (30) day period. If Landlord makes such repairs, Landlord may bill Tenant for the cost of the repairs as additional rent, and said additional rent shall be payable by Tenant within ten (10) days after demand by Landlord. Notwithstanding anything to the contrary contained herein, in the event that any replacement or repair to the Premises constitutes a capital improvement or repair (as determined in accordance with GAAP) (hereinafter, a "Capital Replacement"), and provided, (i) such Capital Replacement is not necessitated by the negligence of Tenant or Tenant's employees, agents or contractors, then such Capital Replacement shall be reimbursed by Tenant as follows: the entire cost of the Capital Replacement shall be amortized over the useful life of the improvement or equipment, as reasonably determined by Landlord, together with an interest factor on the unamortized cost of such item equal to ten percent (10%) per annum and Tenant shall pay such amortized amount monthly during the Term of the Lease (as the same may be extended or renewed) at the same time it pays Base Rent hereunder.

12.2 MAINTENANCE CONTRACTS. Tenant shall, at its own cost and expense, enter into regularly scheduled preventative maintenance/ service contract(s) (the "MAINTENANCE CONTRACTS") with a maintenance contractor for servicing the HVAC units servicing the Premises, the elevator servicing the Premises, the sprinkler, fire alarm and fire detection systems servicing the Premises, and backflow testing for the plumbing servicing the Premises. Tenant shall provide Landlord with copies of all service reports. The service contract must include all services suggested by the equipment manufacturer within the operation/maintenance manual and must become effective (and a copy thereof delivered to Landlord) within thirty (30) days of the date Tenant takes possession of the Premises. Within the ninety (90) day period prior to expiration of the Lease Term, or promptly following other termination of this Lease or of Tenant's right of possession, Landlord shall have the right to have any or all of such systems inspected by a qualified inspector at Landlord's expense. If, in the opinion of the inspector, the equipment has not been properly maintained, reasonable wear and tear excepted, then Landlord may authorize necessary repairs and/or replacements to be made to the system, in which case Tenant shall reimburse Landlord for any and all reasonable costs associated with such inspection and such repairs and replacements. If, at

8

anytime during the Term of the Lease, Landlord discovers that Tenant has failed to enter into or maintain any Maintenance Contract required to be maintained by Tenant hereunder, then Landlord shall have the right, but not the obligation, to enter into such Maintenance Contract and Tenant shall reimburse Landlord for the cost of the Maintenance Contracts within ten (10) days after written demand by Landlord; provided, however, Landlord shall have the right to estimate the monthly cost of the Maintenance Contracts, and Tenant shall pay such amount to Landlord as additional rent each month at the same time Tenant pays Base Rent.

12.3 ADDITIONAL REPAIR OBLIGATIONS. Notwithstanding anything to the contrary in this Article 12, if Tenant leases 100% of the Building, then in addition to Tenant's obligations in Section 12.1 and Section 12.2 above, Tenant shall, at Tenant's sole cost and expense, take good care of and keep the Common Areas of the Project in good and safe order and condition, and shall make all repairs therein and thereon necessary to keep the Common Areas in good and safe order and condition. The term "Common Areas," as used in this Lease, shall mean and include roadways, alleyways, pedestrian sidewalks, driveways, parking lots, truck ways, landscaped areas and those areas within the Project which are not leased or available for lease, or intended to be leased, to a tenant. Common Areas shall not include any of the buildings (including the Building) located at the Project. Tenant shall keep the Common Areas clean and free from dirt, snow, ice, rubbish, obstruction and encumbrances. All repairs made by Tenant shall be equal in quality and class to the original work and shall be made in compliance with all applicable Regulations. As used in this Section 12.3, the term "repairs" shall include all necessary replacements, removals, alterations and additions which are not the responsibility of Landlord under this Lease. Notwithstanding anything to the contrary set forth above, Tenant shall not make any repairs which constitute a Capital Replacement to the Common Areas without Landlord's prior approval. All Capital Replacements made pursuant to this Section 12.3 shall be paid for by Landlord and reimbursed by Tenant in accordance with Section 12.1 hereinabove. If Tenant fails to keep the Common Areas of the Project in good condition and repair, Landlord may, upon thirty (30) days prior written notice to Tenant, perform any necessary maintenance (but shall not be obligated to do so). If Landlord performs any such maintenance obligation, then except as otherwise expressly provided in this Lease, the cost thereof shall be considered an Operating Expense, reimbursable by Tenant in accordance with Article 6.

12.4 PERFORMANCE OF WORK BY LANDLORD. If Tenant is not leasing 100% of the Building, Landlord shall employ contractors to perform all repairs, maintenance and replacements of the sprinkler systems, fire alarm systems, fire detection systems and embedded plumbing systems located within the walls or under the floor of the Premises (the "Building Systems") but only to the extent any of such Building Systems service other space in the Building as well (other than the Premises). The items described in the previous sentence that Landlord will cause to be repaired, maintained and replaced are hereinafter referred to as the "Landlord Maintenance Items.". The cost of all such Landlord Maintenance Items shall be considered an Operating Expense, reimbursable by Tenant in accordance with Article 6. Landlord shall determine in its sole discretion the scope and

timing of the performance of such Landlord Maintenance Items, and Tenant shall not perform such Landlord Maintenance Items. Landlord's maintenance of the exterior walls of the Premises shall include the right, but not the obligation, of Landlord to paint from time to time all or some of the exterior walls, canopies, doors, windows, gutters, handrails and other exterior parts of the Premises with colors selected by Landlord and the cost thereof shall be considered an Operating Expense, reimbursable by Tenant in accordance with Article 6. If the Premises contains landscaped areas ("Landscaped Areas"), Landlord shall maintain the Landscaped Areas in the same condition as they exist as of the date hereof and the cost thereof shall be considered an Operating Expense, reimbursable by Tenant in accordance with Article 6. Tenant shall immediately give Landlord written notice of any repair or maintenance required by Landlord pursuant to this Section, after which Landlord shall have a reasonable time in which to complete such repair or maintenance. Landlord shall have the right at any time, and from time to time, to elect upon written notice to Tenant to have Tenant perform some or all of the Landlord Maintenance Items and/or the maintenance of the Landscaped Areas, in which event Tenant shall employ contractors reasonably designated by Landlord to perform such work and shall pay for all such work at Tenant's sole cost and expense, all in accordance with the requirements of Section 12.1. In addition, from time to time, Tenant may request in writing from Landlord the right to perform all or some of the Landlord Maintenance Items and Landlord shall permit Tenant to perform such Landlord Maintenance Item(s) so long as (i) such Landlord Maintenance Item is not performed to or does not affect the exterior of the Building or any structural element of the Building, and (ii) Tenant employs a contractor reasonably approved by Landlord to perform any such Landlord Maintenance Item. Notwithstanding anything to the contrary contained herein, in the event that any replacement or repair of a Landlord Maintenance Item constitutes a Capital Replacement, and provided, (i) such Capital Replacement is not necessitated by the negligence of Tenant or Tenant's employees, agents or contractors, and (ii) if applicable, Tenant has entered into and maintains a regularly scheduled preventive maintenance/service contract for such system if required by Section 12.2 above, then such Capital Replacement shall be reimbursed by Tenant as follows: the entire cost of the Capital Replacement shall be amortized over the useful life of the improvement or equipment, as reasonably determined by Landlord, together with an interest factor on the unamortized cost of such item equal to ten percent (10%) per annum and Tenant shall pay such amortized amount annually during the Term of the Lease (as the same may be extended or renewed) at the same time it pays Base Rent hereunder.

12.5 Notwithstanding anything to the contrary herein, the parties hereby agree that Tenant shall not be responsible to maintain the Common Areas in accordance with Section 12.3 herein unless and until the first day following the rent abatement period provided in Paragraph 4 of the Addendum with respect to Unit C. Unless such repairs are the direct result of the gross negligence or willful misconduct

9

of Landlord or its employees, agents or contractors, in no event shall Landlord be obligated to reimburse Tenant for any repairs made by Tenant to the Common Areas.

13. ALTERATIONS AND SURRENDER.

13.1 CONSENT OF LANDLORD. Tenant shall not, without Landlord's prior written consent, which may be given or withheld in Landlord's sole discretion, make any alterations, improvements, additions, utility installations or repairs (hereinafter collectively referred to as "ALTERATIONS") in, on or about the Premises or the Project. Landlord shall respond to Tenant's written request for consent hereunder within fifteen (15) business days after Landlord's receipt of such written request. If Landlord fails to respond in writing to Tenant's written request for Landlord's approval of any proposed Alteration (the "First Request") within fifteen (15) business days following Landlord's receipt of the First Request, then Tenant shall send to Landlord a second written request for Landlord's approval (the "Second Request"), which such Second Request shall contain a statement in bold letters of a large font at the top such document stating that "Landlord's failure to respond to this document within seven (7) business days will constitute Landlord's deemed approval of the Alteration requested herein." If Landlord fails to respond to the Second Notice within seven (7) business days following Landlord's receipt of the Second Notice, then such failure shall be deemed to constitute Landlord's approval of the

Alteration. If, as a result of any Alteration made by Tenant, Landlord is obligated to comply with the Americans With Disabilities Act or any other law or regulation, and such compliance requires Landlord to make any improvement or alteration to any portion of the Project, as a condition to Landlord's consent, Landlord shall have the right to require Tenant to pay to Landlord, prior to the construction of any Alteration by Tenant, the entire cost of any improvement or alteration Landlord is obligated to complete by such law or regulation. In performing Alterations, Tenant shall use only such architect and contractor as has been expressly approved by Landlord in the exercise of Landlord's reasonable discretion. In addition, in connection with any Alteration requiring Landlord's consent, Tenant shall pay Landlord's reasonable out-of-pocket costs incurred to review Tenant's Alteration plans within thirty (30) days of Tenant's receipt of an invoice therefor accompanied by reasonably satisfactory supporting documentation. In the event Tenant makes any Alterations, Tenant agrees to obtain or cause its contractor to obtain, prior to the commencement of any work, "builders all risk" insurance in an amount reasonably approved by Landlord, workers' compensation insurance and any other insurance requested by Landlord and consistent with Local Standards. The foregoing notwithstanding, Landlord will not unreasonably withhold, condition or delay its consent to any non-structural Alteration provided that Tenant otherwise complies with the provisions of this Article 13 and that such Alterations (i) are not visible from the exterior of the Premises, and (ii) do not adversely affect any of the Building systems or structure. Furthermore, Tenant shall have the right to make cosmetic, non-structural Alterations (i.e., painting, carpeting, wall papering) to the Premises without obtaining Landlord's prior written consent, provided that Tenant provides Landlord with prior written notice of its intention to make such Alterations. In addition, Tenant shall have the right to make non-structural Alterations that (i) do not adversely affect any of the Building systems or structure, (ii) are located solely within the interior of the Building, and (ii) cost less than Fifty Thousand Dollars (\$50,000.00) per Alteration (hereinafter, a "Minor Alteration") to the Premises without obtaining Landlord's prior written consent, provided that Tenant provides Landlord with prior written notice of its intention to make such Minor Alteration and otherwise complies with the provisions of this Article 13.

- 13.2 PERMITS. Any Alterations in or about the Premises that Tenant shall desire to make shall be presented to Landlord in written form, with plans and specifications which are sufficiently detailed to obtain a building permit. If Landlord consents to an Alteration, the consent shall be deemed conditioned upon Tenant acquiring a building permit from the applicable governmental agencies, furnishing a copy thereof to Landlord prior to the commencement of the work, and compliance by Tenant with all conditions of said permit in a prompt and expeditious manner. Tenant shall provide Landlord with as-built plans and specifications for any Alterations made to the Premises. Tenant shall give Landlord not less than ten (10) days' advance written notice prior to the commencement of any work in the Premises by Tenant, and Landlord shall have the right to post notices of non-responsibility in or on the Premises or the Project.
- 13.3 MECHANICS LIENS. Tenant shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Tenant at or for use in the Premises, which claims are or may be secured by any mechanic's or materialmen's lien against the Premises or the Project, or any interest therein. If Tenant shall, in good faith, contest the validity of any such lien, Tenant shall furnish to Landlord a surety bond reasonably satisfactory to Landlord in an amount equal to the amount of such contested lien claim indemnifying Landlord against liability arising out of such lien or claim. Such bond shall be sufficient in form and amount to free the Project from the effect of such lien. In addition, Landlord may require Tenant to pay Landlord's reasonable attorneys' fees and costs in participating in such action.
- 13.4 SURRENDER. Subject to Landlord's right to require removal or to elect ownership as hereinafter provided, all Alterations made by Tenant to the Premises shall be the property of Tenant, but shall be considered to be a part of the Premises. Unless Landlord gives Tenant written notice of its election not to become the owner of the Alterations at the end of the Term of this Lease, the Alterations shall become the property of Landlord at the end of the Term of this Lease. Landlord may require, on notice to Tenant, that some or all Alterations (other than Cosmetic Alterations) be removed at the end of the Term of this Lease and that any damages caused by such removal be repaired at Tenant's sole expense, provided, however, that to the extent Landlord's consent is required pursuant to this Section, at the written request of Tenant, Landlord agrees to notify Tenant concurrently with Landlord's consent to such Alteration whether or not Landlord will require Tenant to remove such Alteration at the end of the Term. On the last day of the Term hereof, or on any sooner termination, Tenant

Premises to Landlord in substantially the same condition as received by Tenant, ordinary wear and tear and casualty damage excepted, broom-clean and free of Tenant's personal property, trade fixtures and equipment. Tenant's personal property shall include all computer wiring and cabling installed by Tenant. Provided, however, if Landlord has not elected to have Tenant remove the Alterations in accordance with this Section 13.4, Tenant shall leave the Alterations at the Premises in the condition same are required to be maintained by Tenant hereunder, ordinary wear and tear and casualty damage excepted. Tenant shall repair any damage to the Premises occasioned by the installation or removal of Tenant's trade fixtures, furnishings and equipment.

13.5 FAILURE OF TENANT TO REMOVE PROPERTY. If this Lease is terminated due to the expiration of its Term or otherwise, and Tenant fails to remove its personal property, in addition to any other remedies available to Landlord under this Lease, and subject to any other right or remedy Landlord may have under applicable law, Landlord may deem all or any part of Tenant's personal property to be abandoned and title to Tenant's property shall vest in Landlord.

14. DAMAGE AND DESTRUCTION.

14.1 EFFECT OF DAMAGE OR DESTRUCTION. If all or part of the Project is damaged by fire, earthquake, flood, explosion, the elements, riot, the release or existence of Hazardous Materials (as defined below) or by any other cause whatsoever (hereinafter collectively referred to as "DAMAGES"), but the damages are not material (as defined in Section 14.2 below), Landlord shall repair the damages to the Project as soon as is reasonably possible, and this Lease shall remain in full force and effect. If all or a substantial portion of the Project is materially damaged, Landlord shall have the right, in its sole and complete discretion, to repair or to rebuild the Project or to terminate this Lease. Landlord shall within ninety (90) days after the discovery of such material damage notify Tenant in writing of Landlord's intention to repair or to rebuild or to terminate this Lease. Notwithstanding the foregoing, if Landlord shall elect to rebuild or repair the Project after material damage, but in good faith determines that the Premises cannot be substantially repaired within two hundred seventy (270) days after the date of the discovery of the material damage (or within ninety (90) days if the damage occurs during the last eighteen (18) months of the Term of the Lease as the same may be extended), without payment of overtime or other premiums, and the damage to the Project will render all or a material portion of the Premises unusable during said two hundred seventy (270) day period, Landlord shall notify Tenant thereof in writing at the time of Landlord's election to rebuild or repair, and Tenant shall thereafter have a period of fifteen (15) days within which Tenant may elect to terminate the term of this Lease. Tenant's termination right described in the preceding sentence shall not apply if the damage was caused by the negligent or intentional acts of Tenant or its employees, agents or contractors. Failure of Tenant to exercise said election within said fifteen (15) day period shall constitute Tenant's agreement to accept delivery of the Premises under this Lease whenever tendered by Landlord, provided Landlord thereafter promptly pursues reconstruction or restoration diligently to completion, subject to delays caused by Force Majeure Events, as hereinafter defined. If Landlord is unable to repair the damage to the Premises or the Project during such two hundred seventy (270) day period due to Force Majeure Events, the two hundred seventy (270) day period shall be extended by the period of delay caused by the Force Majeure Events.

14.2 DEFINITION OF MATERIAL DAMAGE. Provided that Landlord has maintained the insurance coverage required to be maintained by Landlord under this Lease, damage to the Project shall be deemed material if, in Landlord's reasonable judgment, the uninsured cost of repairing the damage will exceed Five Hundred Thousand Dollars (\$500,000.00). Damage to the Project shall also be deemed material if (a) the Project cannot be rebuilt or repaired to substantially the same condition it was in prior to the damage due to laws or regulations in effect at the time the repairs will be made, (b) the holder of any mortgage or deed of trust encumbering the Property requires that insurance proceeds available to repair the damage in excess of Two Hundred Fifty Thousand Dollars (\$250,000.00) be applied to the repayment of the indebtedness secured by the mortgage or the deed of trust (provided that Landlord agrees to negotiate in good faith, at the time such instrument is being entered into, in order to remove from any mortgage or deed of trust entered into after the date of this

Lease any language which would require Landlord to pay insurance proceeds to the holder of such instrument in the event of a casualty), or (c) fifty percent (50%) or more of the Premises is damaged during the last twelve (12) months of the Term and such damage was not caused by the gross negligence or intentional acts of Landlord or its employees, agents or contractors.

14.3 ABATEMENT OF RENT. If Landlord elects to repair damage to the Project and all or part of the Premises will be unusable or inaccessible to Tenant in the ordinary conduct of its business until the damage is repaired, and the damage was not caused by the negligence or intentional acts of Tenant or its employees, agents or contractors, Tenant's Base Rent, Tenant's Percentage Share of Operating Expenses and Tenant's Percentage Share of Real Property Taxes and Insurance Costs and all other additional rent shall be abated until the repairs are completed in proportion to the amount of the Premises which is unusable or inaccessible to Tenant in the ordinary conduct of its business. Landlord shall not be liable to Tenant or its employees, agents, contractors, invitees or customers for loss or damage to merchandise, tenant improvements, fixtures, automobiles, furniture, equipment, computers, files or other property (hereinafter collectively "TENANT'S PROPERTY") located at the Project unless same arises from the gross negligence or intentional acts of Landlord or its employees, agents or contractors. Except as expressly set forth herein, Tenant shall repair or replace all of Tenant's Property at Tenant's sole cost and expense. In the event of a casualty which results in an abatement of rent pursuant to this Section 14.3, Tenant shall have the option to cease maintaining the Common Areas as more fully provided in Section 12.3 until such time as the Premises are repaired, in which case Landlord shall take over the

11

maintenance of the Common Areas, as more fully set forth in Section 12.4 herein, and the cost thereof will be included in Operating Expenses, which Operating Expenses may be abated as expressly provided in this Section 14.3.

15. CONDEMNATION. If any portion of the Premises or the Project are taken under the power of eminent domain, or sold under the threat of the exercise of said power (all of which are herein called "CONDEMNATION"), this Lease shall terminate as to the part so taken as of the date the condemning authority takes title or possession, whichever first occurs; provided that if so much of the Premises or Project are taken by such condemnation as would substantially and adversely affect the operation of Tenant's business conducted from the Premises, and said taking lasts for sixty (60) days or more, Tenant shall have the option, to be exercised only in writing within thirty (30) days after Landlord shall have given Tenant written notice of such taking (or in the absence of such notice, within thirty (30) days after the condemning authority shall have taken possession), to terminate this Lease as of the date the condemning authority takes such possession. If a taking lasts for less than sixty (60) days, Tenant's rent and additional rent shall be abated during said period but Tenant shall not have the right to terminate this Lease. If Tenant does not terminate this Lease in accordance with the foregoing, this Lease shall remain in full force and effect as to the portion of the Premises remaining, except that the Base Rent, Operating Expenses, Real Property Taxes, Insurance Costs and all other additional rent charges shall be reduced in the proportion that the usable floor area of the Premises taken bears to the total usable floor area of the Premises. Any award for the taking of all or any part of the Premises or the Project under the power of eminent domain or any payment made under threat of the exercise of such power shall be the property of Landlord, whether such award shall be made as compensation for diminution in value of the leasehold, for good will, for the taking of the fee, as severance damages, or as damages for tenant improvements; provided, however, that Tenant shall be entitled to any separate award for loss of or damage to Tenant's removable personal property and for moving expenses.

16. ASSIGNMENT AND SUBLETTING.

16.1 LANDLORD'S CONSENT REQUIRED. Tenant shall not voluntarily or by operation of law assign, transfer, hypothecate, mortgage, sublet, or otherwise transfer or encumber all or any part of Tenant's interest in this Lease or in the Premises (hereinafter collectively a "TRANSFER"), without Landlord's prior written consent, which consent shall not be unreasonably withheld. Landlord shall respond to Tenant's written request for consent hereunder within fifteen (15) business days after Landlord's receipt of a written request from Tenant. If Landlord fails to respond in writing to such written request for Landlord's approval of any proposed Transfer (the "First Request") within fifteen (15)

business days following Landlord's receipt of the First Request, then Tenant shall send to Landlord a second written request for Landlord's approval (the "Second Request"), which such Second Request shall contain a statement in bold letters of a large font at the top such document stating that "Landlord's failure to respond to this document within seven (7) business days will constitute Landlord's deemed approval of the Transfer requested herein." If Landlord fails to respond to the Second Notice within seven (7) business days following Landlord's receipt of the Second Notice, then such failure shall be deemed to constitute Landlord's approval of the Transfer. Any attempted Transfer without such consent shall be void and shall constitute a default under this Lease. If the entity(ies) which directly or indirectly controls the voting shares/rights of Tenant changes at any time, such change of ownership or control shall constitute a Transfer unless Tenant is an entity whose outstanding stock is listed on a recognized securities exchange or if at least 80% of its voting stock is owned by another entity, the voting stock of which is so listed. Tenant's written request for Landlord's consent shall include all of the following information: (a) financial statements for the proposed assignee or subtenant, (b) a reasonably detailed description of the business the assignee or subtenant intends to operate at the Premises, and (c) a copy of the executed letter of intent for the proposed sublease or assignment which includes the material terms and conditions of the proposed assignment or sublease (provided, however, if any of the material terms of such sublease or assignment agreement materially change from the approved letter of intent, Landlord's right to consent and Landlord's fifteen (15) business day period shall once again apply with respect to the revised proposed sublease or assignment).

Notwithstanding anything to the contrary contained in this Article 16, provided Tenant is not in default after expiration of any applicable notice and cure periods, Tenant shall have the right, without Landlord's consent, upon fifteen (15) days advance written notice to Landlord, to assign the Lease or sublet the whole or any part of the Premises (a) to any entity or entities which are owned by Tenant, or which owns Tenant, (b) in connection with the sale or transfer of substantially all of the assets of Tenant or the sale or transfer of substantially all of the outstanding ownership interests in Tenant, or (c) in connection with a merger, consolidation or other corporate reorganization of Tenant (each of the transactions referenced in the above subparagraphs (a), (b), and (c) are hereinafter referred to as a "Permitted Transfer," and each surviving entity shall hereinafter be referred to as a "Permitted Transferee"); provided, that such assignment or sublease is subject to the following conditions:

(i) Tenant shall remain fully liable under the terms of the Lease;

(ii) such Permitted Transfer shall be subject to all of the terms, covenants and conditions of the Lease;

12

(iii) if the entity which constitutes Tenant does not survive such transaction referenced above, such Permitted Transferee has a net worth at least equal to the net worth of Tenant as of the date of this Lease; and

(iv) such Permitted Transferee shall expressly assume the obligations of Tenant under the Lease by a document reasonably satisfactory to Landlord.

16.2 STANDARD FOR APPROVAL. Landlord shall not unreasonably withhold its consent to a Transfer provided that Tenant has complied with the requirements, terms and conditions of this Article 16. It shall be deemed reasonable for Landlord to withhold its consent to a Transfer if the requirements, terms or conditions of this Article 16 are not complied with or: (a) the Transfer would cause Landlord to be in violation of its obligations under another lease or agreement to which Landlord is a party; (b) in Landlord's reasonable judgment, a proposed assignee has a net worth of less than Twenty-Five Million Dollars (\$25,000,000.00) as of the date of such request for Transfer; (c) a proposed assignee's or subtenant's business will impose a burden on the Project's parking facilities, Common Areas or utilities that is substantially greater than the burden imposed by Tenant, in Landlord's reasonable judgment; (d) Tenant is in default as defined in Article 17 at the time of the request; (e) Landlord has sued or been sued by the proposed assignee or subtenant or has otherwise been involved in a legal dispute with the proposed assignee or subtenant; (f) the

proposed assignee or subtenant is an existing tenant of the Project or is a person or entity then negotiating with Landlord for the lease of space in the Project; or (g) the assignee or subtenant will use, store or handle Hazardous Materials in or about the Premises of a type, nature, quantity not acceptable to Landlord, in Landlord's reasonable discretion. In the event Tenant shall assign this Lease or sublet the Premises or request the consent of Landlord to any Transfer, then Tenant shall pay Landlord's reasonable out-of-pocket costs and expenses incurred in connection therewith, including, but not limited to, attorneys', architects', accountants', engineers' or other consultants' fees.

16.3 ADDITIONAL TERMS AND CONDITIONS. Regardless of Landlord's consent, no Transfer shall release Tenant from Tenant's obligations hereunder or alter the primary liability of Tenant to pay the rent and other sums due Landlord hereunder and to perform all other obligations to be performed by Tenant hereunder or release any guarantor from its obligations under its guaranty. The consent by Landlord to any Transfer shall not constitute a consent to any subsequent Transfer by Tenant or to any subsequent or successive Transfer by an assignee or subtenant and no assignment or sublease may be modified or amended without Landlord's prior written consent. However, Landlord may consent to subsequent Transfers or any amendments or modifications thereto without notifying Tenant or anyone else liable on the Lease and without obtaining their consent, and such action shall not relieve such persons from liability under this Lease. In the event of any default under this Lease, Landlord may proceed directly against Tenant, any guarantors or anyone else responsible for the performance of this Lease, including any subtenant or assignee, without first exhausting Landlord's remedies against any other person or entity responsible therefor to Landlord, or any security held by Landlord. The discovery of the fact that any financial statement relied upon by Landlord in giving its consent to an assignment or subletting was materially false shall, at Landlord's election, render Landlord's consent null and void. Any assignee of, or subtenant under, this Lease shall, by reason of accepting such assignment or entering into such sublease, be deemed, for the benefit of Landlord, to have assumed and agreed to conform and comply with each and every term, covenant, condition and obligation herein to be observed or performed by Tenant during the term of said assignment or sublease, other than such obligations as are contrary or inconsistent with provisions of an assignment or sublease to which Landlord has specifically consented in writing.

16.4 TRANSFER PREMIUM FROM ASSIGNMENT OR SUBLETTING. Tenant shall pay Landlord fifty percent (50%) of all rent and other consideration which Tenant receives as a result of a Transfer that is in excess of the rent payable to Landlord for the portion of the Premises and Term covered by the Transfer. Tenant shall pay Landlord for Landlord's share of the excess within thirty (30) days after Tenant's receipt of the excess. Tenant may deduct from the excess all expenses directly incurred by Tenant attributable to the Transfer. If Tenant is in default, Landlord may require that all sublease payments be made directly to Landlord, in which case Tenant shall receive a credit against rent in the amount of Tenant's share of payments received by Landlord. The provisions of this Section 16.4 shall not apply to a Permitted Transfer.

16.5 LANDLORD'S OPTION TO RECAPTURE SPACE. Notwithstanding anything to the contrary contained in this Article 16, in the event Tenant is not leasing 100% of the Building, Landlord shall have the option, by giving written notice to Tenant within thirty (30) days after receipt of any request by Tenant (i) to assign this Lease, or (ii) to sublease space in the Premises if the term of such sublease is scheduled to expire during the last twelve (12) months of the Term of the Lease, to terminate this Lease with respect to said space as of the date thirty (30) days after Landlord's election. In the event of a recapture by Landlord, if this Lease shall be canceled with respect to less than the entire Premises, the Base Rent, Operating Expenses, Real Property Taxes, Insurance Costs and the number of parking spaces Tenant may use shall be adjusted on the basis of the number of rentable square feet retained by Tenant in proportion to the number of rentable square feet contained in the original Premises, and this Lease as so amended shall continue thereafter in full force and effect, and upon request of either party, the parties shall execute written confirmation of same

13

17. DEFAULT; REMEDIES.

17.1 DEFAULT BY TENANT. Landlord and Tenant hereby agree that the occurrence of any one or more of the following events is a default by Tenant

under this Lease and that said default shall give Landlord the rights described in Section 17.2. Landlord or Landlord's authorized agent shall have the right to execute and to deliver any notice of default, notice to pay rent or quit or any other notice Landlord gives Tenant.

- (a) Tenant's failure to make any payment of Base Rent, Operating Expenses, Real Property Taxes, Insurance Costs or any other payment required to be made by Tenant hereunder, as and when due, where such failure shall continue for a period of five (5) days after written notice thereof from Landlord to Tenant.
- (b) The abandonment of the Premises by Tenant, in which event Landlord shall not be obligated to give any notice of default to Tenant. For purposes of this Lease, the Premises shall not be deemed abandoned so long as Tenant (i) has provided Landlord with prior written notice, (ii) pays rent and all other charges due under this Lease as and when due, and (iii) maintains the physical appearance of the Premises.
- (c) Tenant's failure to comply with any of the covenants, conditions or provisions of this Lease to be observed or performed by Tenant (other than those referenced in Sections 17.1(a) and (b) above), where such failure shall continue for a period of thirty (30) days after written notice thereof from Landlord to Tenant; provided, however, that if the nature of Tenant's nonperformance is such that more than thirty (30) days are reasonably required for its cure, then Tenant shall be allowed additional time (not to exceed 90 days) as is reasonably necessary to cure the failure so long as Tenant commences such cure within said thirty (30) day period and thereafter diligently pursues such cure to completion.
- (d) (i) The making by Tenant or any guarantor of Tenant's obligations hereunder of any general assignment for the benefit of creditors; (ii) the appointment of a trustee or receiver to take possession of substantially all of Tenant's assets located at the Premises or of Tenant's interest in this Lease, where possession is not restored to Tenant within sixty (60) days; (iii) the attachment, execution or other judicial seizure of substantially all of Tenant's assets located at the Premises or of Tenant's interest in this Lease, where such seizure is not discharged within sixty (60) days; or (iv) the initiating of an insolvency or bankruptcy proceeding against Tenant, which proceeding is not discharged within sixty (60) days of the filing thereof. In the event that any provision of this Section 17.1(d) is unenforceable under applicable law, such provision shall be of no force or effect.

17.2 REMEDIES.

- (a) In the event of any Tenant default under Section 17.1, Landlord may, at any time thereafter, with or without notice or demand, and without limiting Landlord in the exercise of any right or remedy which Landlord may have by reason of such default:
- (i) Terminate Tenant's right to possession of the Premises. Upon any such termination, Tenant shall immediately surrender possession of the Premises to Landlord. Landlord reserves all rights and remedies available to it pursuant to the terms and conditions of this Lease as well as under applicable law. Tenant hereby grants Landlord the right to enter the Premises with process of law. Tenant releases Landlord of any liability for any damage resulting therefrom and waives any right to claim damage for such re-entry. Tenant also agrees that Landlord's right to re-lease or any other right given to Landlord as a consequence of Tenant's default hereunder or by operation of law is not relinquished. On termination of Tenant's right of possession, Landlord shall be entitled to recover from Tenant: (i) the unpaid rent which had been earned at the time of the termination; (ii) the amount by which the unpaid rent which would have been earned after termination until the time of the award exceeds the amount of rent, if any, received for the Premises during such time period; (iii) the amount by which the unpaid rent for the balance of the Term of the Lease after the time of award exceeds the amount of any rent to be received (net of re-letting expenses as described below) from any replacement tenant occupying the Premises at the time of the award, or, if the Premises are not occupied at the time of the award by a rent-paying replacement tenant, the full amount of the rent to be earned hereunder for the balance of the Term of the Lease discounted to net present value assuming a discount rate of one percent (1%) above the discount rate of the Federal Reserve Bank of Richmond in effect at the time of the award; and provided further, however, that Landlord shall repay to Tenant the excess of the foregoing amount over any rent received for the Premises during the balance of the Term of the Lease (net of re-letting expenses as described below) similarly discounted as and when received; and (iv) at the

time of the award any other amount necessary to compensate Landlord for damages proximately caused by Tenant's failure to perform its obligations under this Lease or which in the ordinary course of events would likely result therefrom, including but not limited to, costs and expenses attributable to recovering possession of the Premises, re-letting expenses (including the costs and expenses of any necessary repairs, renovations and alterations to the Premises), costs of carrying the Premises (including but not limited to, Landlord's payment of real property taxes and insurance premiums), actual legal fees and associated costs and expenses, the unamortized portion of brokerage commissions paid in connection with this Lease and costs of tenant improvements (amortized, without interest, on a straight-line basis

14

over the initial Term of the Lease), and reimbursement of any deferred rent or other Lease execution inducement.

- (ii) maintain Tenant's right of possession, in which event Landlord shall have the remedy which permits Landlord to continue this Lease in effect after Tenant's breach and abandonment and recover rent as it becomes due. In the event Landlord elects to continue this Lease in effect, Tenant shall have the right to sublet the Premises or assign Tenant's interest in the Lease subject to the reasonable requirements contained in Section 16 of this Lease.
- (iii) collect sublease rents (or appoint a receiver to collect such rents) and otherwise perform Tenant's obligations at the Premises, it being agreed, however, that the appointment of a receiver for Tenant shall not constitute an election by Landlord to terminate this Lease.
- (iv) pursue any other remedy now or hereafter available to Landlord under the laws or judicial decisions of the state in which the Premises are located. $\,$
- (b) No remedy or election hereunder shall be deemed exclusive, but shall, wherever possible, be cumulative with all other remedies at law or in equity. The expiration or termination of this Lease and/or the termination of Tenant's right to possession of the Premises shall not relieve Tenant of liability under any indemnity provisions of this Lease as to matters occurring or accruing during the Term of the Lease prior thereto or by reason of Tenant's occupancy of the Premises.
- (c) If Tenant abandons or vacates the Premises and same is a default under Section 17.1 hereof, Landlord may re-enter the Premises, and such re-entry shall not be deemed to constitute Landlord's election to accept a surrender of the Premises or to otherwise relieve Tenant from liability for its breach of this Lease. No surrender of the Premises shall be effective against Landlord unless Landlord has entered into a written agreement with Tenant in which Landlord expressly agrees to (i) accept a surrender of the Premises and (ii) relieve Tenant of liability under the Lease. The delivery by Tenant to Landlord of possession of the Premises shall not constitute the termination of the Lease or the surrender of the Premises.
- 17.3 DEFAULT BY LANDLORD. Except as otherwise expressly provided in this Lease, Landlord shall not be in default under this Lease unless Landlord fails to perform obligations required of Landlord within thirty (30) days after written notice by Tenant to Landlord, specifying wherein Landlord has failed to perform such obligation; provided, however, that if the nature of Landlord's obligation is such that more than thirty (30) days are required for its cure, then Landlord shall not be in default if Landlord commences performance within such thirty (30) day period and thereafter diligently pursues the same to completion.
- 17.4 LATE CHARGES. If any installment of Base Rent, Real Property Taxes, Insurance Costs or any other sum due from Tenant shall not be received by Landlord within five (5) days of when such amount shall be due, then, without any requirement for notice or demand to Tenant, Tenant shall immediately pay to Landlord a late charge equal to six percent (6%) of such overdue amount. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs Landlord will incur by reason of late payment by Tenant. Acceptance of such late charge by Landlord shall in no event constitute a waiver of Tenant's default with respect to such overdue amount, nor prevent Landlord from exercising any of the other rights and remedies granted hereunder,

including the assessment of interest under Section 17.5. Notwithstanding anything to the contrary contained in this Section 17.4 of the Lease, Landlord agrees to waive imposition of such late charge on the first two (2) occasions in any twelve (12) month period provided the overdue payment is made within five (5) business days after Landlord gives Tenant written notice that the payment was not made when due.

- 17.5 INTEREST ON PAST-DUE OBLIGATIONS. Except as expressly herein provided, any amount due to Landlord that is not paid within five (5) days of its due date shall bear interest at the lesser of ten percent (10%) per annum or the maximum rate permitted by applicable law. Payment of such interest shall not excuse or cure any default by Tenant under this Lease.
- 18. LANDLORD'S RIGHT TO CURE DEFAULT; PAYMENTS BY TENANT. In the event of a Tenant default under Section 17.1 hereof, Landlord shall have the right to make any such payment or perform any such act on Tenant's behalf without waiving its rights hereunder and without releasing Tenant from any of its Lease obligations. Tenant shall reimburse Landlord for the cost of such performance upon demand.
- 19. INDEMNITY. Tenant hereby agrees to indemnify, defend and hold harmless Landlord and its employees, partners, agents, contractors, lenders and ground lessors (said persons and entities are hereinafter collectively referred to as the "INDEMNIFIED PARTIES") from and against any and all liability, loss, cost, damage, claims, loss of rents, liens, judgments, penalties, fines, settlement costs, investigation costs, cost of consultants and experts, attorneys' fees, court costs and other legal expenses, effects of environmental contamination, cost of environmental testing, removal, remediation and/or abatement of Hazardous Materials (as said term are defined below), insurance policy deductibles and other expenses (hereinafter collectively referred to as "DAMAGES") arising out of or related to an Indemnified Matter (as defined below). For purposes of this Section, an "INDEMNIFIED MATTER" shall mean any matter for which one or more of the Indemnified Parties incurs liability or Damages if the liability or Damages arise directly out of (a) Tenant's or its employees', agents', contractors' or invitees' (all of said persons or entities are hereinafter collectively referred to as "TENANT PARTIES") use or occupancy of the Premises or the Project,

15

- (b) any act, omission or neglect of a Tenant Party, (c) Tenant's failure to perform any of its obligations under the Lease, (d) the existence, use or disposal of any Hazardous Material brought on to the project by a Tenant Party or (e) any other matters for which Tenant has agreed to indemnify Landlord pursuant to any other provision of this Lease. This indemnity is intended to apply to the fullest extent permitted by applicable law. Tenant's obligations under this Section shall survive the expiration or termination of this Lease unless specifically waived in writing by Landlord after said expiration or termination.
- 20. EXEMPTION OF LANDLORD FROM LIABILITY. Tenant hereby agrees that Landlord shall not be liable for injury to Tenant's business or any loss of income therefrom or for loss of or damage to Tenant's Property, Tenant's employees, agents, contractors or invitees, or any other person in or about the Project, nor shall Landlord be liable for injury to the person of Tenant, Tenant's employees, agents, contractors or invitees, whether such damage or injury is caused by or results from any cause whatsoever including, but not limited to, theft, criminal activity at the Project, negligent security measures, bombings or bomb scares, Hazardous Materials, fire, steam, electricity, gas, water or rain, flooding, breakage of pipes, sprinklers, plumbing, air conditioning or lighting fixtures, or from any other cause, whether said damage or injury results from conditions arising upon the Premises or upon other portions of the Project, or from other sources or places, or from new construction or the repair, alteration or improvement of any part of the Project, unless the cause of the damage or injury arises out of Landlord's or its employees', agents' or contractors' grossly negligent or intentional acts. Landlord shall not be liable for any damages arising from any act or neglect of any employees, agents, contractors or invitees of any other tenant, occupant or user of the Project.
- 21. LIMITATIONS ON LIABILITY. Landlord shall have the right to transfer all or any portion of its interest in the Project and to assign this Lease to the transferee. Upon transfer, Landlord shall automatically be released from all liability under this Lease arising after the date of such transfer; and Tenant

hereby agrees to look solely to Landlord's transferee for the performance of Landlord's obligations hereunder after the date of the transfer. Subject to the rights of any lender holding a mortgage or deed of trust encumbering all or part of the Project, Tenant agrees to look solely to Landlord's equity interest in the Project (including any insurance and/or sales proceeds attributable to the Project) for the collection of any judgment requiring the payment of money by Landlord arising out of (a) Landlord's failure to perform its obligations under this Lease or (b) the negligence or willful misconduct of Landlord, its partners, employees and agents. No other property or assets of Landlord shall be subject to levy, execution or other enforcement procedure for the satisfaction of any judgment or writ obtained by Tenant against Landlord. No partner, shareholder, officer, director, employee or agent of Landlord or Tenant shall be personally liable for the performance of such party's obligations hereunder or be named in any lawsuit arising out of or related to, directly or indirectly, this Lease and the obligations of such party hereunder. The obligations under this Lease do not constitute personal obligations of the individual partners, shareholders, officers, directors or employees of Landlord or Tenant, if any, and neither party shall seek recourse against the assets of any such individuals.

22. SIGNS. Tenant shall not make any changes to the exterior of the Premises, install any exterior lights, decorations, balloons, flags, pennants, banners or painting, or erect or install any signs which can be viewed from the exterior of the Premises, without Landlord's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed. Upon vacating the Premises, Tenant shall remove all signs and repair, paint and/or replace the building facia surface to which its signs are attached. Tenant shall obtain all applicable governmental permits and approvals for signs and exterior treatments.

SEE ADDENDUM PARAGRAPH 8

23. INTENTIONALLY DELETED

24. BROKER'S FEE. Tenant and Landlord each represent and warrant to the other that neither has had any dealings or entered into any agreements with any person, entity, broker or finder other than the persons, if any, listed in Section 1.13, in connection with the negotiation of this Lease, and no other broker, person, or entity is entitled to any commission or finder's fee in connection with the negotiation of this Lease, and Tenant and Landlord each agree to indemnify, defend and hold the other harmless from and against any claims, damages, costs, expenses, attorneys' fees or liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings, actions or agreements of the indemnifying party. Landlord shall pay any commission(s) due the Brokers and Landlord.

25. ESTOPPEL CERTIFICATE. Tenant shall from time to time, upon not less than ten (10) days' prior written notice from Landlord, execute and deliver to Landlord a statement in writing certifying such information with respect to this Lease as Landlord may reasonably request including, but not limited to, the following: (a) that this Lease is unmodified and in full force and effect (or, if modified, stating the nature of such modification and certifying that this Lease, as so modified, is in full force and effect), (b) the date to which the Base Rent and other charges are paid in advance and the amounts so payable, (c) that there are not, to Tenant's knowledge, any uncured defaults or unfulfilled obligations on the part of Landlord, or specifying such defaults or unfulfilled obligations, if any are claimed, (d) that all tenant improvements to be constructed by Landlord, if any, have been completed in accordance with Landlord's obligations, and (e) that Tenant has taken possession of the Premises. Any such statement may be conclusively relied upon by any prospective purchaser or encumbrancer of the Project. Landlord shall at

16

any time, upon not less than ten (10) business days' prior written notice from Tenant, execute, acknowledge and deliver to Tenant a statement in writing certifying to any proposed transferee, or any other person designated by Tenant, as of the date of such estoppel certificate, that (i) this Lease is unmodified and in full force and effect (or if there have been any modifications, that the Lease is in full force and effect as modified and stating the modifications), (ii) the dates to which the rent and any other charges hereunder have been paid,

(iii) whether or not, to the knowledge of Landlord, Tenant is in default in the performance of any covenant, agreement or condition contained in this Lease, and if so, specifying the nature of such default, (iv) the address to which notices to Landlord are to be sent, and (v) such other information with respect to this Lease as Tenant may reasonably request.

26. FINANCIAL INFORMATION. From time to time, at Landlord's request, Tenant shall cause the following financial information to be promptly delivered to Landlord, at Tenant's sole cost and expense: (a) a current financial statement for Tenant and Tenant's financial statements for the previous two accounting years, (b) a current financial statement for any guarantor(s) of this Lease and the guarantor'(s) financial statements for the previous two accounting years and (c) such other financial information pertaining to Tenant or any guarantor as Landlord or any lender or purchaser of Landlord may reasonably request. Notwithstanding the foregoing, so long as Tenant is a publicly traded company, Tenant need only to submit its annual report to comply with the provisions of this Article 26.

27. ENVIRONMENTAL MATTERS/HAZARDOUS MATERIALS.

- (a) For purposes of this Lease, the following additional definitions shall apply:
- (i) "Hazardous Substances" shall include any pollutants, petroleum products, dangerous substances, toxic substances, hazardous wastes, hazardous materials, or hazardous substances as defined in or pursuant to the Industrial Site Recovery Act and all rules, regulations, orders, directives and opinions promulgated thereunder ("ISRA") N.J.S.A. 13:1K-6 et seq.; the Spill Compensation and Control Act, N.J.S.A. 58:10-23.11 et seq. and all rules, regulations, orders, directives and opinions promulgated thereunder ("Spill Act"); the Solid Waste Management Act, N.J.S.A. 13:1E-1 et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. Section. 6901 et seq.; the Comprehensive Environmental Response Compensation and Liability Act, 42 U.S.C. Section. 9601 et seq. and all rules, regulations, orders, directives and opinions promulgated thereunder ("CERCLA"); or any other Federal, State or Local environmental law or ordinance; and all rules, regulations, orders, directives and opinions promulgated under the foregoing, any amendments to any of the foregoing and any successor legislation to any of the foregoing (collectively "Environmental Laws");
- (ii) "Release" means releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, disposing or dumping;
- (iii) "Notice" means any summons, citation, directive, order, claim, litigation, investigation, proceeding, judgment, letter, submission or other communication, written or oral, actual or threatened, from the New Jersey Department of Environmental Protection ("DEP"), the United States Environmental Protection Agency ("EPA"), any other Federal, State or Local agency or authority or any other entity or any individual, concerning any act or omission at the Premises or Project resulting or which may result in the Release of Hazardous Substances into the waters or onto the lands of the State of New Jersey or into waters outside the jurisdiction of the State of New Jersey or into the "environment" as such terms are defined in CERCLA, or otherwise related to any Environmental Law or Tenant's obligations pursuant to this Section 27. "Notice" shall include the imposition of any liens of any real or personal property or revenues of Tenant including, but not limited to, Tenant's interest in the Premises or any of Tenant's property located thereon, pursuant to or resulting from the violation of any Environmental Law or any other governmental actions, orders or permits.
- (b) To the extent that Tenant may be permitted under applicable law to use the Premises and/or the Project for the generating, manufacturing, refining, transporting, treating, storing, handling, disposing, transferring or processing of Hazardous Substances, Tenant shall ensure that said use shall be conducted at all times strictly in accordance with applicable Environmental Law. Tenant shall not cause nor permit as a result of any intentional or unintentional act or omission, a Release of Hazardous Substances at the Premises or the Project. If any intentional or unintentional act or omission results in any actual or alleged Release of Hazardous Substances at the Premises or the Project, Tenant promptly shall conduct necessary sampling and cleanup and remediate such Release in accordance with applicable Environmental Laws.
- (c) Tenant shall not operate any business at the Premises which shall be subject to ISRA. TENANT HEREBY REPRESENTS THAT ITS STANDARD INDUSTRIAL

CLASSIFICATION (HEREIN "SIC") NUMBERS ARE 5136 AND 5137 as determined by reference to the SIC Manual and its operations shall consist of the Use described in Section 1.5. Notwithstanding any provision of ISRA to the contrary, if the Tenant's operations at the Premises become subject to ISRA, Tenant, at Tenant's own expense, shall do whatever is necessary to comply with ISRA whenever an obligation to do so arises. If requested to do so by Landlord, but no more often than once per year, Tenant shall obtain a letter from DEP or any successor agency confirming that ISRA does not apply to Tenant's operations at the Premises. At no expense to Landlord, Tenant promptly shall provide all information requested by Landlord regarding or in furtherance of Tenant's ISRA compliance at the Premises. Tenant shall sign any affidavit concerning compliance with Environmental Laws submitted by Landlord which is true, accurate and complete; if an affidavit is not true,

17

accurate and complete, Tenant shall provide the necessary information to make it true, accurate or complete and then shall sign same.

- (d) Tenant promptly shall furnish Landlord with true copies of any Notices of any nature made by Tenant to, or received by Tenant from DEP, EPA, or any local, state or federal authority with respect to this Project.
- (e) Notwithstanding anything in this Lease to the contrary, and without limiting any other provisions of this Article 27, Tenant, at its sole cost and expense, shall observe, comply and fulfill all of the terms and provisions of all applicable Environmental Laws, as the same may be amended from time to time, as they relate to Tenant's use and occupancy of the Premises during the term of this Lease.

Without limiting the foregoing, Tenant agrees that with respect to its use and occupancy of the Premises, that it shall not do or omit to do nor suffer the commission or omission of any act, the commission or omission of which is prohibited by or may result in liability pursuant to any Environmental Law, including without limitation, the Release of Hazardous Substances.

- (f) In the event there shall be filed a lien against the Premises and/or the Project arising out of a claim(s) by DEP pursuant to the provisions of the Spill Act or by EPA pursuant to the provisions of CERCLA arising from the act or omission of Tenant or its agents, Tenant immediately either shall: 1) pay the claim and remove the lien from the Premises and/or the Project; or, ii) furnish a bond, cash receipt or other security satisfactory to Landlord sufficient to discharge the claim out of which the lien arises.
- (g) (i) Tenant promptly shall provide Landlord with all documentation and correspondence provided to DEP with respect to Tenant's use and occupancy of the Premises pursuant to the Worker and Community Right to Know Act, N.J.S.A. 34:5A-1 et seq., and all rules, regulations, orders, directives and opinions promulgated thereunder.
- (i) Tenant promptly shall supply Landlord all reports and notices made by Tenant with respect to Tenant's use and occupancy of the Premises pursuant to the Hazardous Substance Discharge Reports and Notices Act, N.J.S.A. 13:1K-15, et seq. and all rules, regulations, orders, directives and opinions promulgated thereunder.
- (ii) Tenant promptly shall provide Landlord with a copy of all permits obtained pursuant to any Environmental Law with respect to Tenant's use and occupancy of the Premises.
- (h) Tenant acknowledges that for Landlord to comply with the requirements of Environmental Laws, Landlord from time to time, may have to enter the Premises. Landlord and/or its agents shall have an irrevocable license and right to enter the Premises for such purposes. All such entry by Landlord and/or its agents shall be upon reasonable notice to Tenant. Landlord shall use commercially reasonable efforts not to unreasonably interfere with Tenant's business operations during any such access.
- (i) Tenant agrees to cooperate with Landlord to provide any information necessary to Landlord in order to effect compliance with any Environmental Law and to execute any documents requested by Landlord in connection with compliance with any Environmental Law.

- (j) Tenant shall cooperate fully in allowing, from time to time, such examinations, tests, inspections and reviews of the Premises (hereinafter, the "Tests") as Landlord, in its sole and absolute discretion, shall determine to be advisable in order to evaluate any potential environmental problems or Tenant's compliance with Environmental Laws. Such Tests shall be performed at Landlord's sole expense unless any such Tests determine that Tenant has caused any environmental problems in which event Tenant shall pay for such Tests. Landlord shall use commercially reasonable efforts not to unreasonably interfere with Tenant's business operations during any such access.
- (k) Tenant shall indemnify, defend and hold Landlord harmless from any and all fines, suits, procedures, claims, liabilities, costs and actions of any kind, including counsel fees (including those incurred to enforce this indemnity) arising out of (1) any spills or discharges of Hazardous Substances at the Premises and/or Project for which Tenant is responsible in accordance with the terms of this Lease and (2) Tenant's failure to comply with the provisions of this Article 27. Tenant's obligations and liabilities pursuant to this indemnity shall continue for so long as Landlord remains responsible or liable under Environmental Laws or otherwise for any spills or discharges of Hazardous Substances and/or for any violations of Environmental Laws which have occurred as a result of the acts or omissions of Tenant at the Project. Tenant's failure to abide by the terms of this Article shall be enforceable by injunction.
- (1) Notwithstanding anything to the contrary contained in this Lease, Tenant shall not be responsible for complying with any Environmental Law with respect to the Project and shall have no liability or responsibility whatsoever under this Lease or otherwise to Landlord in connection with any

18

spill, discharge or Release of Hazardous Substances at the Project occurring prior to the Commencement Dates of this Lease or which were not caused by the act or omission of Tenant or its agents.

- (m) In no event shall Landlord be liable or responsible to Tenant or anyone claiming through or under Tenant for the failure of any other tenant or other person to comply with any Environmental Law.
- (n) In the event Tenant shall fail to comply in full with this Article, Landlord, at its option, may perform any and all of Tenant's obligations as aforesaid, and all costs and expenses so incurred by Landlord shall be deemed a claim against Tenant as Additional Rent payable on demand.
- (o) The provisions of this Article 27 shall survive the expiration or earlier termination of this Lease, regardless of the reason for such termination and compliance with the provisions of this Article 27 may require Tenant to expend funds or perform acts after the expiration or termination of this Lease. Tenant agrees to expend such funds and/or perform such acts and shall not be excused therefrom notwithstanding any expiration or termination of this Lease, it being agreed and acknowledged that Landlord would not have entered into this Lease but for the provisions of this Article 27.
- (p) Landlord represents and warrants to the best of its knowledge and belief that, as of the date of this Lease, there are no Hazardous Materials on, in or under the Premises or the Building in violation of any applicable Environmental Laws. Landlord shall indemnify, defend and hold harmless Tenant from and against any and all claims, judgments, damages, penalties, fines, costs, liabilities or losses which arise as a result of the existence or disposal of Hazardous Materials brought into the Building by Landlord, its employees, contractors or agents violation of any Environmental Laws. Landlord's foregoing indemnity shall survive the expiration or earlier termination of the Lease.
- 28. SUBORDINATION. This Lease shall be subject and subordinate to any ground lease, mortgage, deed of trust or any other hypothecation or security now or hereafter placed upon the Project and to any and all advances made on the security thereof and to all renewals, modifications, consolidations, replacements and extensions thereof (collectively referred to as a "MORTGAGE"). Notwithstanding such subordination, Tenant's right to quiet possession of the Premises shall not be disturbed if Tenant is not in default under this Lease

after expiration of all applicable notice and cure periods. At the request of any mortgagee, trustee or ground lessor, Tenant shall attorn to such person or entity. As an alternative, a mortgagee, trustee or ground lessor shall have the right at any time to subordinate its Mortgage to this Lease. Tenant agrees to execute and acknowledge any documents Landlord reasonably requests Tenant execute to effectuate an attornment, a subordination, or to make this Lease prior to the lien of any Mortgage.

29. INTENTIONALLY DELETED

- 30. HOLDING OVER. If Tenant remains in possession of the Premises or any part thereof after the expiration or earlier termination of the Term hereof with Landlord's consent, such occupancy shall be a tenancy from month to month upon all the terms and conditions of this Lease pertaining to the obligations of Tenant, except that the Base Rent payable shall be one hundred fifty percent (150%) of the Base Rent payable immediately preceding the termination date of this Lease, calculated on a per diem basis, and all options, if any, shall be deemed terminated and be of no further effect. If Tenant remains in possession of the Premises or any part thereof after the expiration of the Term hereof without Landlord's consent, Tenant shall be treated as a tenant at sufferance or a trespasser. Nothing contained herein shall be construed to constitute Landlord's consent to Tenant holding over at the expiration or earlier termination of the Term or to give Tenant the right to hold over after the expiration or earlier termination of the Term. If Tenant remains in possession of the Premises or any part thereof for a period in excess of sixty (60) days after the expiration or earlier termination of the Term hereof, Tenant hereby agrees to indemnify, hold harmless and defend Landlord from any cost, loss, claim or liability (including attorneys' fees) Landlord may incur as a direct result of Tenant's failure to surrender possession of the Premises to Landlord upon the termination of this Lease.
- 31. LANDLORD'S ACCESS. Landlord and Landlord's agents, contractors and employees shall have the right to enter the Premises at reasonable times and upon twenty-four (24) hours notice (except in the event of an emergency where no such notice shall be required) for the purpose of inspecting the Premises, performing any services required of Landlord, showing the Premises to prospective purchasers, lenders or, during the last nine (9) months of the Term, tenants, undertaking safety measures and making alterations, repairs, improvements or additions to the Premises or to the Project. In the event of an emergency, Landlord may gain access to the Premises by any reasonable means, and Landlord shall not be liable to Tenant for damage to the Premises or to Tenant's property resulting from such access. If reasonably necessary, Landlord may temporarily limit or eliminate access to portions of the Project, or perform work in the Building, which work may create noise, dust or leave debris in the Building. The foregoing activities by Landlord shall not constitute constructive eviction or entitle Tenant to an abatement or reduction of rent provided that Landlord shall use commercially reasonable efforts not to unreasonably interfere with Tenant's business operations during any such access and shall promptly complete any work undertaken by Landlord. Landlord shall have the right to retain keys to the locks on the entry doors to the Premises and all interior doors at the Premises.

19

32. MISCELLANEOUS.

- 32.1 SEVERABILITY. The invalidity of any provision of this Lease as determined by a court of competent jurisdiction shall in no way affect the validity of any other provision hereof.
- 32.2 DEFINITION OF ADDITIONAL RENT. All monetary obligations of Tenant to Landlord under this Lease, including, but not limited to, Base Rent, Operating Expenses, Real Property Taxes, Insurance Costs and late charges shall be deemed to be rent.
- 32.3 INCORPORATION OF PRIOR AGREEMENTS. This Lease constitutes the entire agreement between the parties and supersedes all prior agreements and understandings related to the Premises, including all lease proposals, letters of intent and other documents. Neither party is relying upon any warranty, statement or representation not contained in this Lease.
- 32.4 WAIVERS. No waiver by Landlord or Tenant of any provision hereof shall be deemed a waiver of any other provision hereof or of any subsequent breach by

Landlord or Tenant of the same or any other provision. Landlord's consent to, or approval of, any act shall not be deemed to render unnecessary the obtaining of Landlord's consent to or approval of any subsequent act by Tenant. The acceptance of rent hereunder by Landlord shall not be a waiver of any preceding breach by Tenant of any provision hereof, other than the failure of Tenant to pay the particular rent so accepted, regardless of Landlord's knowledge of such preceding breach at the time of acceptance of such rent. The payment of rent hereunder by Tenant shall not be a waiver of any preceding breach by Landlord of any provision hereof regardless of Tenant's knowledge of such preceding breach at the time of Tenant's payment of such rent. No acceptance by Landlord of partial payment of any sum due from Tenant shall be deemed a waiver by Landlord of its right to receive the full amount due, nor shall any endorsement or statement on any check or accompanying letter from Tenant be deemed an accord and satisfaction. Tenant hereby waives for Tenant and all those claiming under Tenant all rights now or hereafter existing to redeem by order or judgment of any court or by legal process or writ, Tenant's right of occupancy of the Premises after any termination of this Lease.

- 32.5 SECURITY MEASURES. Tenant hereby acknowledges that Landlord shall have no obligation whatsoever to provide guard service or other security measures for the benefit of the Premises or the Project, and Landlord shall have no liability to Tenant due to its failure to provide such services. Tenant assumes all responsibility for the protection of Tenant, its agents, employees, contractors and invitees and the property of Tenant and of Tenant's agents, employees, contractors and invitees from acts of third parties.
- 32.6 AMENDMENTS. This Lease may be modified in writing only, signed by the parties in interest at the time of the modification.
- 32.7 FORCE MAJEURE EVENTS. Whenever a period of time is prescribed for the taking of an action by Landlord or Tenant (other than the payment of the Security Deposit or rent), the period of time for the performance of such action shall be extended by the number of days that the performance is actually delayed due to strikes, acts of God, shortages of materials, war, terrorist acts, civil disturbances and other causes beyond the reasonable control of the performing party ("FORCE MAJEURE EVENTS").
- 32.8 BINDING EFFECT; CHOICE OF LAW; CONFLICT. Subject to any provision hereof restricting assignment or subletting by Tenant, this Lease shall bind the parties, their heirs, personal representatives, successors and assigns. This Lease shall be governed by the laws of the state in which the Project is located and any litigation concerning this Lease between the parties hereto shall be initiated in the county in which the Project is located.
- 32.9 ATTORNEYS' FEES. If Landlord or Tenant brings an action to enforce the terms hereof or declare rights hereunder, the prevailing party in any such action, or appeal thereon, shall be entitled to its reasonable attorneys' fees and court costs to be paid by the losing party as fixed by the court in the same or separate suit, and whether or not such action is pursued to decision or judgment. Landlord shall be entitled to reasonable attorneys' fees and all other costs and expenses incurred in the preparation and service of notices of monetary default provided that at least one (1) monetary default shall have occurred during the preceding twelve (12) month period whether or not a legal action is subsequently commenced in connection with such default. Landlord and Tenant agree that attorneys' fees incurred with respect to defaults and bankruptcy are actual pecuniary losses within the meaning of Section 365(b) (1) (B) of the Bankruptcy Code or any successor statute.
- 32.10 MERGER; RELATIONSHIP OF PARTIES. The voluntary or other surrender of this Lease by Tenant, or a mutual cancellation thereof, or a termination by Landlord, shall not result in the merger of Landlord's and Tenant's estates, and shall, at the option of Landlord, terminate all or any existing subtenancies or may, at the option of Landlord, operate as an assignment to Landlord of any or all of such subtenancies. Nothing contained in this Lease shall be deemed or construed by the parties hereto or by any third party to create the relationship of principal and agent, partnership, joint venturer or any association between Landlord and Tenant.

20

cure periods, Tenant shall have quiet possession of the Premises for the entire Term hereof.

- 32.12 AUTHORITY; MULTIPLE PARTIES. Tenant represents and warrants to Landlord that each individual executing this Lease on behalf of Tenant is authorized to do so on behalf of Tenant and that this Lease is enforceable against said entity in accordance with its terms. If more than one person or entity is named as Tenant herein, the obligations of Tenant shall be the joint and several responsibility of all persons or entities named herein as Tenant. Service of a notice in accordance with Article 34 on one Tenant shall be deemed service of notice on all Tenants. Landlord represents and warrants to Tenant that each individual executing this Lease on behalf of Landlord is authorized to do so on behalf of Landlord and that this Lease is enforceable against said entity in accordance with its terms.
- 32.13 INTERPRETATION. This Lease shall be interpreted as if it was prepared by both parties and ambiguities shall not be resolved in favor of Tenant because all or a portion of this Lease was prepared by Landlord. The captions contained in this Lease are for convenience only and shall not be deemed to limit or alter the meaning of this Lease. As used in this Lease, the words tenant and landlord include the plural as well as the singular. Words used in the neuter gender include the masculine and feminine gender. The deletion of any printed, typed or other portion of this Lease shall not evidence the parties' intention to contradict such deleted portion. Such deleted portion shall be deemed not to have been inserted in this Lease.
- 32.14 PROHIBITION AGAINST RECORDING. Neither this Lease, nor any memorandum, affidavit or other writing with respect thereto, shall be recorded by Tenant or by anyone acting through, under or on behalf of Tenant. Landlord shall have the right to record a memorandum of this Lease, and Tenant shall execute, acknowledge and deliver to Landlord for recording any memorandum prepared by Landlord.
- 32.15 SECURITY INTEREST. In consideration of the covenants and agreements contained herein, and as a material consideration to Landlord for entering into this Lease, Tenant hereby unconditionally grants to Landlord a continuing security interest in and to all personal property of Tenant located or left at the Premises and the Security Deposit, if any, and any advance rent payment or other deposit, now in or hereafter delivered to or coming into the possession, custody or control of Landlord, by or for the account of Tenant, together with any increase in profits or proceeds from such property. Notwithstanding anything to the contrary contained in this Section 32.15 of this Lease, the security interest granted by Tenant to Landlord shall be automatically subordinated to the security interest, if any, granted to Tenant's lenders in the ordinary course of Tenant's business. Landlord shall, simultaneously with the execution of the Lease, execute the lien waiver, in the form attached hereto as Exhibit F, waiving Landlord's security interest in the collateral described in any such lien waiver (which collateral shall exclude the Improvements and any fixtures installed in the Premises).
- 32.16 ATTACHMENTS. The items listed in Section 1.15 are a part of this Lease and are incorporated herein by this reference.

SEE ADDENDUM PARAGRAPHS 9, 10 AND 11

- 33. OFAC CERTIFICATION. Tenant certifies that: (i) it is not acting, directly or indirectly, for or on behalf of any person, group, entity, or nation named by any Executive Order or the United States Treasury Department as a terrorist, "Specially Designated National and Blocked Person," or other banned or blocked person, entity, nation, or transaction pursuant to any law, order, rule, or regulation that is enforced or administered by the Office of Foreign Assets Control; and (ii) it is not engaged in this transaction, directly or indirectly on behalf of, or instigating or facilitating this transaction, directly or indirectly on behalf of, any such person, group, entity, or nation. Tenant hereby agrees to defend, indemnify, and hold harmless Landlord from and against any and all claims, damages, losses, risks, liabilities, and expenses (including attorney's fees and costs) arising from or related to any breach of the foregoing certification.
- 34. NOTICES. All notices required or permitted by this Lease shall be in writing and shall be delivered (a) by hand, (b) by U.S. Postal Service certified mail, return receipt requested, (c) by U.S. Postal Service Express Mail, Federal Express or other overnight courier, or (d) by facsimile transmission, and shall be deemed sufficiently given if served in a manner specified in this Section. Any notice hereunder shall be deemed personally delivered to Tenant on the date

the notice is personally delivered to any employee of Tenant at the Premises. The addresses set forth in Section 1.14 of this Lease shall be the address of each party for notice purposes. Landlord or Tenant may by written notice to the other specify a different address for notice purposes. Any notice sent by certified mail, return receipt requested, shall be deemed given three (3) business days after deposited with the U.S. Postal Service. Notices delivered by overnight courier service shall be deemed given on the date delivered by the carrier to the appropriate party's address for notice purposes. If any notice is transmitted by facsimile transmission, the notice shall be deemed delivered upon telephone confirmation of receipt of the transmission thereof. A copy of all notices delivered to a party by facsimile transmission shall also be mailed to the party on the date the facsimile transmission is completed. If Tenant has vacated the Premises or any other notice address of Tenant without providing a new notice address, such notice shall be deemed to have been received three (3) business days after notice is deposited in the U.S. mail or with a courier service in the manner described above. If notice is received on Saturday, Sunday or a legal holiday, it shall be deemed

2.1

received on the next business day. Nothing contained herein shall be construed to limit Landlord's right to serve any notice to pay rent or quit or similar notice by any method permitted by applicable law, and any such notice shall be effective if served in accordance with any method permitted by applicable law whether or not the requirements of this section have been met.

35. WAIVER OF JURY TRIAL. LANDLORD AND TENANT HEREBY WAIVE THEIR RESPECTIVE RIGHT TO TRIAL BY JURY OF ANY CAUSE OF ACTION, CLAIM, COUNTERCLAIM OR CROSS-COMPLAINT IN ANY ACTION, PROCEEDING AND/OR HEARING BROUGHT BY EITHER LANDLORD AGAINST TENANT OR TENANT AGAINST LANDLORD ON ANY MATTER WHATSOEVER ARISING OUT OF, OR IN ANY WAY CONNECTED WITH, THIS LEASE, THE RELATIONSHIP OF LANDLORD AND TENANT, TENANT'S USE OR OCCUPANCY OF THE PREMISES, OR ANY CLAIM OF INJURY OR DAMAGE, OR THE ENFORCEMENT OF ANY REMEDY UNDER ANY LAW, STATUTE, OR REGULATION, EMERGENCY OR OTHERWISE, NOW OR HEREAFTER IN EFFECT.

PREPARATION OF THIS LEASE BY LANDLORD OR LANDLORD'S AGENT AND SUBMISSION OF SAME TO TENANT SHALL NOT BE DEEMED AN OFFER BY LANDLORD TO LEASE THE PREMISES TO TENANT OR THE GRANT OF AN OPTION TO TENANT TO LEASE THE PREMISES. THIS LEASE SHALL BECOME BINDING UPON LANDLORD ONLY WHEN FULLY EXECUTED BY BOTH PARTIES AND WHEN LANDLORD HAS DELIVERED A FULLY EXECUTED ORIGINAL OF THIS LEASE TO TENANT.

LANDLORD TENANT

THE REALTY ASSOCIATES FUND VI, L.P., G-III APPAREL GROUP LTD., a Delaware limited partnership

a Delaware corporation

By: Realty Associates Fund VI LLC, a Massachusetts limited liability company, general partner

By: /s/ Wayne S. Miller _____

Its: Chief Operating Officer

By: Realty Associates Advisors LLC, a Delaware limited liability company, manager

> By: Realty Associates Advisors Trust, a Massachusetts business trust, sole member

> > By: /s/ Christopher J. Good -----Officer, Regional Director

ADDENDUM TO STANDARD INDUSTRIAL LEASE (THE "LEASE"),
DATED THE 29TH DAY OF JUNE, 2006, BETWEEN
THE REALTY ASSOCIATES FUND VI, L.P. ("LANDLORD") AND
G-III APPAREL GROUP LTD. ("TENANT")

It is hereby agreed by Landlord and Tenant that the provisions of this Addendum are a part of the Lease. If there is a conflict between the terms and conditions of this Addendum and the terms and conditions of the Lease, the terms and conditions of this Addendum shall control. Capitalized terms in this Addendum shall have the same meaning as capitalized terms in the Lease, and, if a Work Letter Agreement is attached to this Lease, as those terms have been defined in the Work Letter Agreement.

1 LANDLORD'S WORK.

- (a) Landlord shall, at Landlord's sole cost and expense, on or before the applicable Commencement Dates (it being agreed that Landlord shall only be obligated to complete that portion of Landlord's Work which affects the portion of the Premises being delivered to Tenant by the applicable Commencement Date) (i) repaint and carpet/VCT tile offices throughout the Premises, (ii) create four (4) additional openings (2 on each demising wall) of the same size of the existing opening between spaces, (iii) repair the warehouse floor, where necessary, (iv) deliver all the mechanical systems in good working order, and (v) deliver the Premises in "broom swept" condition, and as more particularly described on Exhibit A attached hereto (collectively, the "Landlord's Work"). Landlord shall complete all of Landlord's Work in a good and workmanlike manner. Except as expressly provided in this Paragraph 1, Landlord shall have no obligation to make any improvements to the Premises on behalf of Tenant.
- (b) Landlord shall, at Landlord's sole cost and expense, perform the repairs to the roof of the Building outlined in Exhibit G, attached hereto. Such work shall be completed on or before December 31, 2006 as the same may be extended for any Force Majuere Events.
- 2 TENANT'S INITIAL WORK. Tenant shall have the right to construct certain improvements to the Premises based on plans and specifications (the "Plans") approved by Landlord in accordance with all of the terms and conditions of the Lease (except that Section 13.1 of the Lease shall not apply with respect to the Tenant's Initial Work defined hereinafter, except as otherwise expressly provided below). Landlord's right to approve the Plans shall include, without limitation, the right to reasonably approve the method of installation and the location of the Tenant's Initial Work. The Landlord approval procedure set forth in Section 13.1 of the Lease shall govern the time periods within which Landlord must respond to Tenant's written requests for Landlord's approval of Tenant's Initial Work Plans. The "Tenant's Initial Work" shall consist of those items set forth on Schedule 1 attached hereto and made a part hereof. Tenant's Plans shall include such items and information as Landlord shall reasonably require to evaluate the Tenant's Initial Work. Tenant shall use the Plans to obtain all permits and approvals which are necessary to construct the Tenant's Initial Work. All of the Tenant's Initial Work shall be constructed by a contractor or contractors reasonably approved by Landlord in a good and workmanlike manner and in accordance with all applicable laws, codes and regulations, including the Americans with Disabilities Act ("ADA"), and using materials reasonably satisfactory to Landlord. It is expressly agreed that (i) Tenant shall not commence any such work until said Plans have been approved or deemed approved by Landlord, and (ii) the Plans which have been so approved or deemed approved by Landlord shall be used by Tenant to obtain all permits that are necessary to construct the Tenant's Initial Work. Notwithstanding anything to the contrary contained in the Lease, Tenant shall be required to remove the Tenant's Initial Work upon the expiration or earlier termination of the Lease Term.
- 3 ROOF-TOP ACCESS. Landlord hereby agrees that Tenant shall have the right of access to and use of (subject to Landlord's right to access for required repairs and inspections to be made by Landlord or its contractors) a portion of the Building roof for Tenant to install, maintain and operate one or more satellite dish(es) or antenna(e) (including appropriate conduit and utilities for the operation thereof), the location of which shall be reasonably agreed upon by Landlord and Tenant, such use by Tenant to be free of charge during the Term hereof. In the event Tenant wishes to place communication equipment on the roof, it shall be (i) screened in a manner and design acceptable to Landlord in its sole but reasonable discretion, (ii) installed and maintained in compliance with applicable codes, and (iii) installed and maintained at Tenant's sole cost and expense. Prior to installing any such equipment on the roof of the Building, Tenant shall execute a license agreement substantially in the form attached to

this Lease as Exhibit E. The rights granted to Tenant pursuant to this Paragraph 3 are personal to Tenant, its Permitted Transferees and any assignees or subtenants of Tenant which have been approved by Landlord (if such consent is required by Section 16 of the Lease) and may not be assigned separate and a part from this Lease.

4 RENT ABATEMENT. Notwithstanding anything to the contrary contained in this Lease, (i) no Base Rent, Operating Expenses, Real Property Taxes or Insurance Costs shall be paid by Tenant in connection with Unit B for the first four (4) consecutive full calendar months of the Term of this Lease following the Unit B Commencement Date, (ii) no Base Rent, Operating Expenses, Real Property Taxes or Insurance Costs shall be paid by Tenant in connection with Unit A for the first four (4) consecutive full calendar months of the Term of this Lease following the Unit A Commencement Date, and (iii) no Base Rent, Operating Expenses, Real Property Taxes or Insurance Costs shall be paid by Tenant in connection with Unit C for the first five (5) consecutive full calendar months of the Term of this Lease

following the Unit C Commencement Date. No other amounts due to Landlord under this Lease, other than the rent referenced above, shall be abated, except as expressly provided in the Lease. In the event Tenant defaults hereunder and fails to cure such default within any applicable notice or cure period, Tenant shall not be entitled to any further abatement of rent and, if such default occurs on or before February 1, 2008, all rent previously abated shall be immediately paid by Tenant to Landlord. This provision shall not relieve Tenant of its obligation to pay the first month's Base Rent set forth in Section 1.9 upon Tenant's execution of this Lease. Such amount shall be applied to Base Rent for the first full calendar month of the Term of the Lease for which Base Rent is due for each of Unit B, Unit A and Unit C.

5 BASE RENT INCREASES. The Base Rent set forth in Section 1.8 above shall be adjusted during the Term of the Lease as follows:

Unit B

Lease Period	Rate Per Square Foot	Monthly Base Rent
10/1/06-9/30/08*	\$4.00	\$33,203.67
10/1/08-9/30/11	\$4.25**	\$35,278.90
10/1/11-1/31/14	\$4.50**	\$37,354.13

- * Does not reflect Base Rent abatement described in Paragraph 4 of the Addendum
- ** The amount set forth herein as the rate per square foot shall be decreased by \$0.05 per square foot to \$4.20 per square foot and \$4.45 per square foot, respectively, in the event Tenant does not exercise its option to terminate pursuant to Paragraph 10 below.

Unit A

Lease Period	Rate Per Square Foot	Monthly Base Rent
01/01/07-12/31/08*	\$4.00	\$35,201.00
01/01/09-12/31/11	\$4.25**	\$37,401.06
01/01/12-1/31/14	\$4.50**	\$39,601.13

- * Does not reflect Base Rent abatement described in Paragraph 4 of the Addendum
- ** The amount set forth herein as the rate per square foot shall be decreased by \$0.05 per square foot to \$4.20 per square foot and \$4.45 per square foot, respectively, in the event Tenant does not exercise its option to terminate pursuant to Paragraph 10 below.

Unit C

Rate Per Square Foot	Monthly Base Rent
\$4.00	\$33,421.00
\$4.20	\$35,092.05
\$4.45	\$37,180.86
	\$4.00

- * Does not reflect Base Rent abatement described in Paragraph 4 of the Addendum
- 6 TENANT'S PERCENTAGE SHARE. Notwithstanding anything to the contrary contained in Sections 1.10 of the Lease, Tenant's Percentage Share shall be adjusted during the Term of the Lease according to the following schedule:

Lease Period

Tenant's Percentage Share*

Unit B Commencement Date through midnight of the day preceding the Unit A Commencement Date

Unit A Commencement Date through midnight of the day preceding the Unit C Commencement Date

Unit C Commencement Date

Unit C Commencement Date through the Unit C Commencement Date

Unit C Commencement Date through the expiration of the Term

Tenant's Percentage Share*

32.61% (based on 99,611 sq.ft)

- 7 SECURITY DEPOSIT. Article 7 of the Lease is hereby amended by adding the following at the end of Article 7:
- (a) The Security Deposit shall be in the form of an irrevocable, unconditional letter of credit (the "Security Deposit L/C") in the amount set forth in Section 1.11, as security for Tenant's full and faithful performance of Tenant's obligations hereunder. The Security Deposit L/C shall be delivered to Landlord at Tenant's sole cost and expense. The Security Deposit L/C shall be issued by and drawn on a bank reasonably acceptable to Landlord, in Landlord's sole discretion, and shall name Landlord as Beneficiary. The CIT Group/Commercial Services, Inc. is hereby approved by Landlord as issuer of the

Security Deposit L/C. The Security Deposit L/C shall be substantially in the form attached hereto as Exhibit D. If the maturity date of the Security Deposit L/C is prior to the end of the Term of the Lease, Tenant shall renew the Security Deposit L/C as often as is necessary with the same bank or financial institution (or a similar bank or financial institution reasonably acceptable to Landlord) and upon the same terms and conditions, not less than thirty (30) days prior to the purported expiration date of the Security Deposit L/C. In the event that Tenant fails to timely renew the Security Deposit L/C as aforesaid, Landlord shall be entitled to draw against the entire amount of the Security Deposit L/C. The Security Deposit L/C shall be assignable by Landlord and upon such assignment to any party assuming in writing the lessor interest in this Lease, Landlord shall be relieved from all liability to Tenant therefor.

- (b) In the event that Tenant defaults under this Lease beyond applicable notice and cure periods, Landlord shall be entitled to draw upon so much of the Security Deposit L/C as equals the defaulted payment(s) or Landlord's actual damages incurred with respect to such default, plus any interest or other charges due thereon in accordance with this Lease. If Landlord elects to make a partial draw upon the Security Deposit L/C, Tenant shall promptly restore the Security Deposit L/C to its original amount within ten (10) days after written demand therefor. Landlord's election to make a partial draw upon the Security Deposit L/C shall in no event prejudice or waive Landlord's right to terminate this Lease if permitted under applicable provisions of this Lease, nor shall such election prejudice or waive any other remedy of Landlord reserved under the terms of this Lease, including the right to draw the entire amount of the Security Deposit L/C, if applicable. The Security Deposit L/Cshall be available for payment against the presentation of a sight draft by the Landlord together with a certificate from Landlord that Tenant is in default of its obligations hereunder beyond expiration of any applicable notice and cure periods and that Landlord is entitled, by the terms of this Lease, to draw upon the Security Deposit L/C. Landlord agrees to copy Tenant on any notice to the issuing bank requesting a draw against the Security Deposit L/C. The proceeds of the Security Deposit L/C, if drawn by Landlord pursuant to the terms hereof, shall be held by Landlord in accordance with the provisions of Section 7 of the Lease and applied to reduce any amount owed by Tenant to Landlord. No interest shall be payable for any Security Deposit L/C proceeds held on account.
- (c) In the event that (1) Landlord draws the full amount of the Security Deposit L/C as a result of a default by Tenant, (2) this Lease is not terminated by Landlord as a result of such default, (3) such default is fully cured by Tenant, and (4) there is no outstanding uncured default by Tenant, then

the balance of the sums drawn (after the payment of any sums related to the curing of any defaults) shall be applied first to obtain a replacement letter of credit as security for Tenant's performance hereunder, and the remaining balance, if any, will be refunded to Tenant. Upon the termination of this Lease and the payment in full to Landlord of all damages, costs and expenses to which Landlord is entitled hereunder, the balance of any funds drawn from the Security Deposit L/C after satisfying such obligations shall be refunded to Tenant.

- (d) To the extent that the Security Deposit L/C is either lost or the issuing bank will not honor the Security Deposit L/C, Tenant will be responsible to promptly replace the amount of the Security Deposit either in cash (to be held in accordance with this Article 7 of the Lease) or with another Letter of Credit meeting the requirements herein.
- 8 EXTERIOR SIGNAGE. Notwithstanding anything to the contrary contained in Article 22 of the Lease, Tenant, at its sole cost, and subject to all applicable codes and regulations, including applicable zoning requirements, and Landlord's reasonable signage and design criteria, and otherwise subject to Landlord's prior approval, not to be unreasonably withheld, conditioned or delayed, including but not limited to, Landlord's approval of the size, location and installation, shall be permitted to place exterior signage on the Building. In the event Tenant does not lease 100% of the Building, the signage rights granted to Tenant in this paragraph are personal to the original Tenant, any Permitted Transferee and any assignee or a subtenant of the entire Premises consented to by Landlord pursuant to Article 16, and Tenant shall not transfer, assign, or otherwise convey its signage rights separate and apart from this Lease.

9 OPTION TO RENEW.

- (a) Provided that Tenant is not in default at the time of Tenant's exercise of the option contemplated by this Paragraph 9 (the "Option") or at the commencement of the Option term, Tenant shall have one (1) five (5) year option to renew this Lease. Tenant shall provide to Landlord on a date which is prior to the date that the Option period would commence (if exercised) by at least two hundred seventy (270) days, a written notice of the exercise of such Option to extend the Lease for the additional Option term, time being of the essence. Such notice shall be given in accordance with Article 34 of the Lease. If notification of the exercise of the Option is not so given, the Option shall automatically expire. Base Rent applicable to the Premises for the renewal term shall be equal to the "Fair Market Rental" (defined hereinafter). All other terms and conditions of the Lease shall remain the same.
- (b) If Tenant exercises the Option, Landlord shall determine the Fair Market Rental by using its good faith judgment. Landlord shall provide Tenant with written notice of such amount within fifteen (15) days after Tenant exercises its Option. Tenant shall have fifteen (15) days ("Tenant's Review Period") after receipt of Landlord's notice of the new base rent within which to accept such rental. In the event Tenant fails to accept in writing such rental proposal by Landlord, then such proposal shall be deemed rejected and Landlord and Tenant shall attempt to agree upon such Fair Market Rental using

good faith efforts. If Landlord and Tenant fail to reach agreement within fifteen (15) days following the expiration of Tenant's Review Period ("Outside Agreement Date") then the parties shall each within ten (10) days following the Outside Agreement Date appoint a real estate broker who shall be licensed in the State of New Jersey and who specializes in the field of commercial, industrial space leasing in the South Brunswick region of New Jersey, has at least five (5) years of experience and is recognized within the field as being reputable and ethical. If one party does not timely appoint a broker, then the broker appointed by the other party shall promptly appoint a broker for such party. Such two individuals shall each determine within ten (10) days after their appointment such Fair Market Rental. If such individuals do not agree on Fair Market Rental, then the two individuals shall, within fifteen (15) days thereafter, render separate written reports of their determinations and together appoint a third similarly qualified individual having the qualifications described above. In the event Landlord's and Tenant's respective brokers are unable to agree upon a third broker, the third broker shall be appointed by the Chief Judge of the Circuit Court of Middlesex County, New Jersey. The third individual shall within ten (10) days after his or her appointment make a determination of such Fair Market Rental. The third individual shall determine which of the determinations of the first two individuals is closest to his own and the determination that is closest shall be final and binding upon the parties, and such determination may be enforced in any court of competent

jurisdiction. Landlord and Tenant shall each bear the cost of its broker and shall share equally the cost of the third broker. Upon determination of the base rent payable pursuant to this Section, the parties shall promptly execute an amendment to this Lease stating the rent so determined.

- (c) The term "Fair Market Rental" shall mean the annual amount per rentable square foot that a willing, comparable renewal tenant would pay and a willing, comparable landlord of a similar building would accept at arm's length for similar space, giving appropriate consideration to the following matters: (i) annual rental rates per rentable square foot; (ii) the type of escalation clauses (including, without limitation, operating expenses, real estate taxes, and CPI) and the extent of liability under the escalation clauses (i.e., whether determined on a "net lease" basis or by increases over a particular base year or base dollar amount); (iii) rent abatement provisions reflecting free rent and/or no rent during the lease term; (iv) length of lease term; (v) size and location of premises being leased; and (vi) other generally applicable terms and conditions of tenancy for similar space; provided, however, Tenant shall not be entitled to any tenant improvement or refurbishment allowance unless specifically negotiated between the parties. The Fair Market Rental may also designate periodic rental increases, a new base year and similar economic adjustments.
- (d) The Option granted to Tenant in this Lease is personal to the original Tenant and may be exercised only by the original Tenant while occupying the entire Premises and may not be exercised or be assigned, voluntarily or involuntarily, by or to any person or entity other than Tenant and any Permitted Transferee. The Options, if any, herein granted to Tenant are not assignable separate and apart from this Lease, nor may any option be separated from this Lease in any manner, either by reservation or otherwise. In the event Tenant does not lease 100% of the Building, then in the event the Lease has been assigned or a sublease exists as to any portion of the Premises, the Option shall be deemed null and void and neither Tenant nor any assignee or subtenant shall have the right to exercise the Option.
- 10 TENANT'S OPTION TO TERMINATE. Tenant shall have the option to terminate this Lease with respect to Unit C only (the "Unit C Termination") by delivering notice to Landlord of its election to terminate the Lease with respect to Unit C on or before October 31, 2006, time being of the essence (the "Termination Date"). If notification of the exercise of the Unit C Termination is not so given and received, the Unit C Termination granted hereunder shall automatically expire and be of no further force and effect. If Tenant exercises the Unit C Termination, (i) Tenant's Percentage Share shall be 67.18%, and (ii) Tenant shall, from and after the Termination Date, have the non-exclusive right to use its pro rata share of parking spaces at the Project. Landlord shall, at Landlord's sole cost and expense, separately demise Units A and B from Unit C, to Tenant's reasonable satisfaction, promptly following the Termination Date. In the event Tenant does not exercise the option to terminate the Lease with respect to Unit C or otherwise the option to terminate lapses and is of no further force and effect, Paragraph 5 of the Addendum to this Lease shall be modified by providing that the rent per square foot to be paid by Tenant for Unit A and Unit B shall each be decreased by \$0.05 per square foot during the second and third rental periods as noted in Paragraph 5.
- 11 CONSEQUENTIAL AND PUNITIVE DAMAGES. Except with respect to damages incurred in connection with Articles 27 and 30 of the Lease, neither party shall be liable to the other for consequential or punitive damages.

EXHIBIT 31.1

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Morris Goldfarb, certify that:

- I have reviewed this quarterly report on Form 10-Q of G-III Apparel Group, Ltd.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: September 13, 2006

/s/ Morris Goldfarb

Morris Goldfarb Chief Executive Officer

EXHIBIT 31.2

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Neal S. Nackman, certify that:

- I have reviewed this quarterly report on Form 10-Q of G-III Apparel Group, Ltd.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared.
 - b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting, and
- The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: September 13, 2006

Neal S. Nackman Chief Financial Officer

EXHIBIT 32.1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of G-III Apparel Group, Ltd. (the "Company") on Form 10-Q for the quarterly period ended July 31, 2006, as filed with the Securities and Exchange Commission (the "Report"), I, Morris Goldfarb, Chief Executive Officer of the Company, hereby certify that, to my knowledge, (a) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and (b) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Morris Goldfarb
-----Morris Goldfarb
Chief Executive Officer

Date: September 13, 2006

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

EXHIBIT 32.2

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of G-III Apparel Group, Ltd. (the "Company") on Form 10-Q for the quarterly period ended July 31, 2006, as filed with the Securities and Exchange Commission (the "Report"), I, Neal S. Nackman, Chief Financial Officer of the Company, hereby certify that, to my knowledge, (a) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and (b) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Neal S. Nackman
----Neal S. Nackman
Chief Financial Officer

Date: September 13, 2006

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.